Direction des Études et Synthèses Économiques

# G 2013 / 12

# Natural Disasters: Exposure and Underinsurance

Céline **GRISLAIN-LETRÉMY** 

Document de travail



Institut National de la Statistique et des Études Économiques

# INSTITUT NATIONAL DE LA STATISTIQUE ET DES ÉTUDES ÉCONOMIQUES

*Série des documents de travail de la Direction des Études et Synthèses Économiques* 

# G 2013 / 12

## Natural Disasters: Exposure and Underinsurance

Céline **GRISLAIN-LETRÉMY**\*

10 OCTOBRE 2013

L'auteur remercie Pierre-André Chiappori, Pierre-Philippe Combes, Xavier D'Haultfoeuille, Eric Dubois, Meglena Jeleva, Guy Laroque, Claire Lelarge, David Martimort, Philippe Mongin, Corinne Prost, Bernard Salanié, Sandrine Spaeter, Eric Strobl, Yoshihiko Suzawa, et Bertrand Villeneuve pour leurs remarques et suggestions. L'auteur remercie Stephen Coate pour son éclairage sur les aides publiques. Ce travail a également bénéficié de commentaires de Lucie Calvet, Nicolas Grislain, Laurence Rioux, Corentin Trevien, Lionel Wilner et des participants à l'école d'été internationale de Belpasso sur l'économie des ressources et de l'environnement de 2011, au séminaire interne du laboratoire de microéconométrie du CREST, au séminaire de microéconomie théorie appliquée de Columbia University, à la 74ème conférence économique atlantique internationale, à la 20ème conférence annuelle de l'association européenne de l'économie des ressources et de l'environnement, au séminaire de la Chaire « Régulation et Risque Systémique » de l'Autorité de Contrôle Prudentiel et au 40ème séminaire du groupe européen des économistes du risques et de l'assurance. L'auteur remercie enfin Francis Roux et Christophe Michel pour leurs explications détaillées sur les données.

Département des Études Économiques - Timbre G201 - 15, bd Gabriel Péri - BP 100 - 92244 MALAKOFF CEDEX - France - Tél. : 33 (1) 41 17 60 68 - Fax : 33 (1) 41 17 60 45 - CEDEX - E-mail : <u>d3e-dg@insee.fr</u> - Site Web Insee : http://www.insee.fr

Ces documents de travail ne reflètent pas la position de l'Insee et n'engagent que leurs auteurs. Working papers do not reflect the position of INSEE but only their author's views.

<sup>\*</sup> Département des Études Économiques - Division « Marchés et entreprises »Timbre G230 - 15, bd Gabriel Péri - BP 100 -92244 MALAKOFF CEDEX

## Natural Disasters: Exposure and Underinsurance

### Abstract

Insurance coverage against natural disasters remains low in many exposed areas. Limited insurance supply is commonly identified as a primary factor causing low insurance coverage. The French overseas departments provide an unusual combination of a well-developed natural disasters insurance supply in highly exposed regions. Indeed, the French natural disasters insurance system is guaranteed by the French government; first foreseen for continental France only, it was extended to overseas departments after Hurricane Hugo in 1989, in a state of emergency. This situation enables to analyze the determinants of insurance coverage on the demand side. Using unique household-level micro-data, I estimate a semi-structural model of insurance market which had not been empirically tested. The structural approach enables to show that underinsurance in the French overseas departments is neither due to perception biases nor to unaffordable insurance, but mainly to uninsurable housing and to anticipated assistance, which crowds out insurance. Individual insurance decision is impacted by neighbors' insurance choices via peer effects and via neighborhood eligibility for assistance.

Keywords: natural disasters, insurance, disaster aid, public assistance

### Catastrophes naturelles : exposition et sous-assurance

### Résumé

La couverture assurantielle contre les catastrophes naturelles reste faible dans de nombreux pays pourtant fortement exposés, comme en Amérique Latine ou dans les Caraïbes. Une raison souvent invoquée pour expliquer cette faible couverture est la limitation de l'offre d'assurance. Les départements d'Outre-mer français présentent une situation rare d'offre d'assurance développée dans ces régions. En effet, le régime français d'assurance contre les catastrophes naturelles est garanti par l'Etat ; il était initialement prévu pour s'appliquer uniquement en métropole et a été étendu aux départements d'Outre-mer dans l'urgence, après l'ouragan Hugo en 1989. Cette situation permet d'analyser les déterminants de la demande d'assurance. Avec une base unique de données microéconomiques relatives aux assurés et aux non-assurés, l'estime un modèle théorique d'équilibre sur le marché de l'assurance habitation, assurance incluant obligatoirement la couverture des catastrophes naturelles. Cette approche structurelle me permet de mesurer les distorsions induites sur la tarification du risque de catastrophes naturelles et de montrer que la faible pénétration de l'assurance est due, non pas au prix de l'assurance ou aux biais de perception, mais au caractère inassurable de certains logements et à l'anticipation par les ménages d'une aide financière, qui se substitue à l'assurance. Je montre également que les décisions d'assurance de leurs voisins modifient les choix d'assurance des ménages, à la fois par des effets de pair et par l'éligibilité commune du guartier à des aides potentielles.

**Mots-clés :** catastrophes naturelles, assurance, aide aux victimes de catastrophes, aides publiques

Classification JEL : Q54, G22, H84, D12

# 1 Introduction

Natural disasters have had an important and growing impact on individual economies; over the last decades, associated damages have frequently reached several percents of GDP.<sup>1</sup> Up to now, the increasing cost of natural disasters is largely explained by the growing urbanization of risky areas (Barredo (2009), Bevere et al. (2011)). In the future, climate change could have a major additional impact (IPCC, 2007). Among the different strategies developed to manage natural risks, insurance as a coverage solution has taken a growing importance over the last thirty years. There is a macroeconomic value of risk transfer to insurance markets, since this transfer greatly facilitates economic recovery. Actually, the national output decrease subsequent to natural disasters is mainly driven by the uninsured losses (von Peter et al., 2012). As government is potentially the "insurer of last resort" after natural disasters, insurance coverage of public and private assets would enable countries to partially transfer catastrophic risk to private foreign actors via insurance mechanisms.<sup>2</sup>

However, risk transfer to insurance markets remains limited. Even if insured losses have significantly increased over time, they still represent a small fraction of economic losses (MunichRe, 2012). Indeed, insurance coverage remains low not only for public goods but also for firms' and households' possessions, even in developed countries.<sup>3</sup> In many developing countries and small island developing states, concurrence of exposure and underinsurance is striking (Cavallo and Noy (2009), Freeman et al. (2003), Pelling and Uitto (2001)). In particular, Latin America and the Caribbean are one of the more disaster-prone areas of the world (Borensztein et al. (2009), Heger et al. (2008), Rasmussen (2004)) and have the lowest levels of insurance cov-

<sup>&</sup>lt;sup>1</sup>Natural Disasters. Counting the Cost. March 21st, 2011. The Economist.

<sup>&</sup>lt;sup>2</sup>In almost all developing countries, insurers heavily rely on international reinsurance (Outreville, 2000). Local insurance companies can cede an important part of their risks to reinsurers, which are mainly foreign companies. For example, in the Caribbean, local insurers which cover households' and firms' possessions against natural disasters retain less than 20% of the amount they insure and cede the remaining portion to reinsurers (Pollner, 2000).

<sup>&</sup>lt;sup>3</sup>For example, insurance coverage is low in the United States (Dixon et al. (2006), Kunreuther (1984)) or in many European countries (Maccaferri et al., 2012)).

erage (Borensztein et al., 2009).

Limited insurance supply is commonly identified as a primary factor causing low insurance coverage in hazard-prone regions of the world, including Latin America and the Caribbean. However the French overseas departments provide a rare natural experiment of a well-developed and regulated natural disasters insurance supply in Latin America, the Caribbean and other exposed small island countries.<sup>4</sup> The French natural disasters insurance system is guaranteed by the French government. This system was created in 1982 and was first foreseen to apply to continental land only. However, following Hurricane Hugo that devastated Guadeloupe in 1989, the government decided, in a state of emergency, to extend the natural disasters insurance system to the French overseas departments. This wide and regulated coverage supply enables to analyze the determinants of insurance coverage on the demand side, some of them being specific to developing countries, some of them also widely applying in developed countries.

The first and main contribution of this paper is to provide explanations on the demand side for underinsurance in disaster-prone areas and to measure and compare their magnitude. A structural approach enables to disentangle the different possible causes of underinsurance in the French overseas departments. I show that the two standard explanations, perception biases and insurance affordability, are precluded.<sup>5</sup> Actually, the two main explanations for the low insurance penetration rate are uninsurable housing and charity hazard. Uninsurable housing, namely the fact that dwellings are so little resilient to natural events that they can be considered as uninsurable by insurers, widely applies in Latin America, the Caribbean and many other developing countries. The impact of uninsurable housing on insurance demand, which is captured by using proxies for low-quality dwellings, is quanti-

<sup>&</sup>lt;sup>4</sup>The French overseas departments include French Guiana (South America), Guadeloupe (Caribbean Sea), Martinique (Caribbean Sea) and Réunion (Indian Ocean). Mayotte (Indian Ocean) became a French overseas department in March 2011. As data were collected in 2006 in the French overseas departments, Mayotte is excluded from this empirical analysis.

<sup>&</sup>lt;sup>5</sup>A companion paper draws first basic and robust qualitative conclusions (Calvet and Grislain-Letrémy, 2011).

fied and important. Charity hazard, that is the fact that assistance is a substitute for formal insurance and decreases demand for insurance, is a typical example of Samaritan's dilemma and concerns many developed and developing countries. The impact of charity hazard is shown thanks to the structural estimation, data on assistance being unavailable, and is also of important magnitude. Besides, I show that the neighbors' insurance choices impact the individual insurance decision via peer effects and via neighborhood eligibility for assistance, these two channels having comparable orders of magnitude. These results contribute to the growing literature about charity hazard (Petrolia et al. (2012), Landry and Jahan-Parvar (2011), Raschky and Weck-Hannemann (2007), Raschky et al. (2010)). Finally, I show that the existing insurance obligations (*de facto* for homeowners with outstanding loans, as in most Caribbean countries, and *de jure* for French tenants) are operant but do not guarantee that targeted households are insured, as households may not renew their insurance contracts once they have settled in.

The second contribution of this paper is to measure the impact of regulation on insurers' pricing behavior. The French government provides an unlimited guarantee to one reinsurer and regulates in return the scope and the price of natural disasters coverage. Beyond strict regulation, the attractive and unactuarial reinsurance policies offered by this reinsurer provide limited incentives for insurers to price natural risks in their insurance policies. Similar pricing distortions have been observed in other markets, for example in the retail electricity market (Joskow and Tirole, 2006) where intermediaries' pricing reflects their limited exposure and not the real price. Besides, as reinsurance policies limit insurers' exposure to natural risks, insurers also have limited incentives to acquire detailed information on their insured risk exposure (*ex ante* moral hazard) and to precisely assess damages (*ex post* moral hazard).

The third and last contribution consists in specifying and estimating a theoretical model of insurance derived from Abel (1986)'s, Pauly (1974)'s and Rothschild and Stiglitz (1976)'s works and which had not been previously tested. In this model, a

supply equation explains the insurance premium; a demand equation explains the probability of purchasing insurance and takes into account the impact of insurance price on the decision to purchase insurance. Such an estimation of demand and supply has been performed on other markets such as the French labor market (Laroque and Salanié, 2002) but is new for an insurance market. A unique household-level micro-database combining detailed information about the insured and the uninsured has been built to estimate this model.

The paper is organized as follows. Section 2 illustrates exposure and underinsurance in Latin America and the Caribbean, presents commonly identified reasons on the insurance supply and demand sides and details the natural disasters insurance supply provided in the French overseas departments. Section 3 presents the theoretical model. Section 4 details the data and the empirical specification, identification and calibration of the model. Estimation results are commented in Section 5. Section 6 discusses the main reasons for uninsurance, that are uninsurable housing and charity hazard, and to what extent they apply in other developing and developed countries. Section 7 concludes.

# 2 Exposure and underinsurance in Latin America and the Caribbean

In many developing countries (Cavallo and Noy (2009), Freeman et al. (2003)) and small island developing states (Pelling and Uitto, 2001), concurrence of exposure and underinsurance is striking. Especially, Latin America and the Caribbean are one of the more disaster-prone areas of the world (Borensztein et al. (2009), Heger et al. (2008), Rasmussen (2004)) and have suffered damages exceeding 50% of GDP (Table 1), while having the lowest levels of insurance coverage (Borensztein et al., 2009): less than 4% of losses were insured between 1985 and 1999, ranking them last among the regions of the world, with Asia (4%) and after Africa (9%) (Charvériat, 2000).<sup>6</sup> The insurance penetration rate, i.e. the percentage of insured economic agents, is particularly low among households (Charvériat, 2000). For example, in Mexico in 1998, less than 1% of houses had disaster insurance coverage (Kreimer et al., 1999); in Argentina, Ecuador and Brazil flood insurance penetration rate is also very low among individuals (Gaschen et al., 1998).

Country	Time	Event	Damages (% of GDP)
St Lucia	1988	Hurricane Gilbert	365
Grenada	2004	Hurricane Ivan	203
Dominica	1979	Hurricanes David and Fredrick	101
St Kitts and Nevis	1995	Hurricane Luis	85
St Lucia	1980	Hurricane Allen	66
Antigua and Barbuda	1995	Hurricane Luis	61
Guyana	2005	Floods	59
	Notes	: Heger et al. (2008).	

Table 1: Destructive impact of natural disasters in the Caribbean region

### 2.1 Insurance supply

### 2.1.1 Limited insurance supply

Limited insurance supply is commonly identified as a primary factor causing low insurance coverage in hazard-prone regions of the world. Insurance supply is particularly limited in developing countries; microinsurance provides increasing but still partial coverage of loss to life, property and crop caused by natural disasters (see Mechler et al. (2006) for a review).<sup>7</sup> Restricted supply is mainly due to unavailable or unaffordable reinsurance and also to limited standardized information on risk exposure (Cavallo and Noy, 2009).

The case of Latin America and the Caribbean is again particularly striking. Coverage supply for governmental expenditures remains limited despite recent advances

<sup>&</sup>lt;sup>6</sup>For example, in 1999, in the cases of the Vargas tragedy in Venezuela and of Quindio earthquake in Colombia, only 1.4% and 4.4% of total losses were insured, respectively (Charvériat, 2000).

<sup>&</sup>lt;sup>7</sup>See also Barnett et al. (2008) for a review of index-based risk transfer products to cover natural damages to crops.

such as the creation of Caribbean Catastrophe Risk Insurance Facility in 2006 or the success of the Mexican government in the issuance of catastrophe bonds in 2006 (Borensztein et al., 2009) and in 2009 (WB, 2011).<sup>8</sup> Similarly, developments of insurance supply for households remain isolated,<sup>9</sup> and this insurance supply can be fragile. The example of Montserrat is particularly telling: in 1997, after several volcanic eruptions, insurance companies responsible for most policies entirely withdrew from the island (Analytica, 1997). Even when available, insurance premiums offered to households are high in Latin America and the Caribbean, because of the limited reinsurance supply (Auffret (2003), Charvériat (2000) and Evans (1996)).

### 2.1.2 The exception of the French overseas departments

The French overseas departments provide a rare natural experiment of a welldeveloped and regulated natural disasters insurance supply in Latin America, the Caribbean and other exposed small island countries.

As many countries located in the same areas, the French overseas departments are highly exposed to tsunamis, floods and ground movements;<sup>10</sup> Guadeloupe and Martinique are exposed to intense seismic activity;<sup>11</sup> each of the three islands is made of one active volcano (Grande Soufrière in Guadeloupe, Mount Pelée in Martinique, Piton de la Fournaise in Réunion) and is exposed to strong hurricanes or cyclones.<sup>12</sup>

<sup>&</sup>lt;sup>8</sup>In the 1980s and the 1990s, Mexico, Colombia, Costa Rica, Nicaragua have also set up national natural disasters funds for uninsured regional or local infrastructures (Charvériat, 2000).

<sup>&</sup>lt;sup>9</sup>In Brazil, the government-owned reinsurance institute largely contributes to develop the supply of floods reinsurance (Charvériat, 2000); in Puerto Rico, a reserve for catastrophe losses was created in 1994 to improve the availability and the affordability of catastrophe insurance (Charvériat (2000), Evans (1996)); in Manizales (Colombia), the city allows any resident to buy insurance to a private insurer through the municipal tax collection system (Fay and Wellenstein, 2005).

 $<sup>^{10}</sup>$ Ground movements include all soil and subsoil movements (such as mudslides, rock and/or block falls, land collapses or subsidence, landslides, movements due to clayey soils).

<sup>&</sup>lt;sup>11</sup>See the French earthquake map: http://www.planseisme.fr/IMG/jpg/Poster\_alea\_ sismique\_avril\_2008-2.jpg. Major earthquakes occurred in Guadeloupe in 1843 and in Martinique in 1839. Earthquakes of smaller intensity can more frequently happen, such as Les Saintes (Guadeloupe) earthquake on November 21, 2004 and Martinique earthquake on November 29, 2007. According to scientists, a major earthquake is foreseen in each of these two islands in the very next decades.

<sup>&</sup>lt;sup>12</sup>Hurricane Dean damaged Guadeloupe and Martinique on August 16, 2007; Cyclone Dina occurred in Réunion on January 22 and 23, 2002.

This is why collective prevention against natural disasters is highly developed in the French overseas departments.<sup>13</sup>

The French overseas departments were integrated into France as overseas departments in 1946 and are now integral parts of France. The French natural disasters insurance system was created in 1982 to institutionalize and coordinate numerous aid mechanisms that had lasted for centuries (Favier and Larhra, 2007). It applied first to continental land only; a specific insurance system was initially foreseen for the overseas departments. However, following Hurricane Hugo that devastated Guadeloupe in 1989, the government decided, in a state of emergency, to extend the natural disasters insurance system to the French overseas departments (Bidan, 2000).<sup>14</sup> Thus, since August 1, 1990 the French overseas departments have benefited from a well-developed and regulated supply of natural disasters insurance. Indeed, the government provides an unlimited guarantee to the French natural disasters insurance system and regulates in return the scope and the price of natural disasters coverage. Thus, the insurance system corresponds to a tax system: the government ultimately compensates insured damages caused by natural disasters and taxes the insured in return.<sup>15</sup>

**Definition of natural disasters.** Natural disasters are defined by the law as uninsurable natural hazards.<sup>16</sup> They can be earthquakes, volcanic eruptions, hurricanes, cyclones, tsunamis, floods or ground movements. In practice, after a natural event, the French government decides whether this event is a natural disaster and

<sup>&</sup>lt;sup>13</sup>The very wide majority of municipalities have already undertaken or set up natural risks prevention plans. In Guadeloupe and Martinique, the government has set up additional measures in 2007 to improve seismic resilience to activity, especially the one of public infrastructure.

<sup>&</sup>lt;sup>14</sup>At that time, the natural disasters insurance system was also extended to two self-governing territorial overseas collectivities of France, Mayotte and Saint Pierre and Miquelon. Mayotte became a French overseas department in March 2011 and so was not a department in 2006, when data were collected.

<sup>&</sup>lt;sup>15</sup>The natural disasters insurance system also provides coverage to firms and local governments. <sup>16</sup>Natural disasters are defined by the law as "uninsurable natural hazards mainly caused by abnormal intensity of a natural agent, when usual measures to prevent from these damages could not prevent their occurring or could not have been taken" (Insurance Code, section L. 125-1).

which periods and municipalities are concerned.<sup>17</sup> The decision relies on the conclusions of an interministerial commission, which analyzes the phenomenon on the basis of scientific reports. Storms (but neither hurricanes nor cyclones) and fire forests are considered as insurable risks; their coverage is not *de jure* but *de facto* included in home insurance and is not regulated.

Insured households. The coverage of dwellings against natural disasters is mandatorily included in comprehensive home insurance,<sup>18</sup> and this coverage is not provided by any other insurance policy to my knowledge. Insurers are not allowed to sell home insurance without this coverage, which guarantees that insurers do not select their clients. Similarly, households are not allowed to buy home insurance without this coverage. Recall that this system was first foreseen to apply to continental France only, where almost all households purchase home insurance. Thus, this mandatory inclusion initially guaranteed a large mutualization of natural risks over the country.

In practice, French insurers offer to households a coverage of their dwelling against several hazards (such as theft, fire, explosion, water damages or natural disasters), without letting them choose their insured value of the building; households can choose their insured value of furniture only.

**Insurance pricing.** The law imposes the natural disasters premium to be a fixed share of the home insurance premium: the premium for natural disasters amounts to 12% of the premium charged for other risks.<sup>19</sup> Insurers are allowed to increase the home insurance premium (and so the natural disasters premium) with respect to the exposure to natural risks.

<sup>&</sup>lt;sup>17</sup>An order of the ministries of the Interior, of the Economy and of the Budget establishes whether an event is a natural disaster and determines the concerned periods, municipalities and hazards covered by the insurance system. Insured households and firms can benefit from the insurance compensation only if an order is published for the considered event.

<sup>&</sup>lt;sup>18</sup>See Insurance Code, section L. 125-1. Home insurance policy is an accessible product, as it can be purchased by households on the phone in approximately 20 minutes.

<sup>&</sup>lt;sup>19</sup>See Insurance Code, sections L. 125-2 and A. 125-2. The premium for natural disasters equals 12% of the premium charged for other damages only, excluding for example the premium of the coverage for civil liability. For the sake of simplicity, the model ignores this point (Section 3). See http://www.ccr.fr for more details.

**Reinsurance policy.** However, using reinsurance policies, the government provides limited incentives for insurers to price natural risks. Indeed, the government provides an unlimited guarantee to one reinsurer, the Caisse Centrale de Réassurance (CCR), which offers to insurers an attractive and unactuarial reinsurance policy and captures more than 90% of the market shares on the natural disasters reinsurance market.<sup>20</sup> Insurers transfer to CCR their natural risks (at the exception of a fixed deductible which equals the sum of their collected premiums); in counterpart, they pay to CCR a fixed share of their collected natural disasters premiums. As potential loss and reinsurance premium paid by the insurer depend on the exposure of his policyholders to a very limited extent, insurance premiums partially reflect risk exposure and insurers have limited incentives to acquire detailed information on their insured risk exposure (*ex ante* moral hazard) and to precisely assess damages (*ex post* moral hazard).

More precisely, the reinsurance policy offered by CCR is such that the insurer yields 50% of the sum of all the natural disasters premiums he has collected (over all policies) and 50% of his losses caused by natural disasters (over all policies) to CCR (quota-share contract).<sup>21</sup> So, the insurer keeps half of the premiums and covers half of the risks. On his remaining risks, he is exposed until a deductible, which equals the sum of the initially collected premiums (stop-loss contract) (Figure 1).<sup>22</sup> In 2006, the amount paid by insurers to CCR corresponded to 51.5% of the collected premiums (Mn €670 over Bn €1.3, Letrémy (2009)), that is 50% as the price of the quota-share policy and 1.5% as the price of the stop-loss policy. In practice, the stop-loss price depends on the composition of the insurer portfolio in terms of professional risks and not household risks.<sup>23</sup>

Finally, insurers also have to give to the French government 12% of the collected

<sup>&</sup>lt;sup>20</sup>Private communication to the author.

<sup>&</sup>lt;sup>21</sup>Since 2000, insurers are not allowed to select which risks they cede to CCR (Erhard-Cassegrain et al., 2006).

 $<sup>^{22}</sup>$ Each year the deductible is reassessed given the provision realized by the insurer.

 $<sup>^{23}</sup>$ Private communication to the author.

premiums to fund prevention measures.<sup>24</sup> Thus, over the initially collected natural disasters premium, the insurer pays 63.5% of the premium, that is 51.5% to CCR and 12% to the government; in counterpart, the insurer is exposed until a deductible which equals the sum of the collected premiums.



Figure 1: Reinsurance policy

This reinsurance policy is globally applied to all natural disasters policies offered by the insurer (home, firm and car insurance in continental land, overseas departments and territories). For the sake of simplicity, the theoretical model (Section 3) compares the premium of one home insurance policy with the additional expected coverage that it represents.

**Insurance penetration rate.** Despite the supply of a wide coverage against natural disasters at a regulated price, in 2006 only half of households living in the French overseas departments have purchased home insurance, which includes coverage against natural disasters, for their main home (Table 2). This penetration rate is higher than the rate observed in other exposed countries, but remains much lower than the rate observed in continental France, where households are far less exposed to natural risks but almost all insured (Table 2).<sup>25</sup>

<sup>&</sup>lt;sup>24</sup>See Environment Code, section L. 561-3.

<sup>&</sup>lt;sup>25</sup>Indeed, we do not observe the adverse selection effect that would be typically expected: insurance subsidization for exposed households by the least exposed ones (living in continental France) could lead to a higher participation on the insurance market of exposed households (living in overseas departments). This adverse selection would mainly come not from the lack of information by insurers but from their limited incentive to use information because of reinsurance policies: as insurers bear a very limited part of losses caused by natural disasters, insurance premiums partially reflect natural risks and this way subsidies exposed households.

	French Guiana	Guadeloupe	e Martinique	Réunion	Continental France
Population					
1	$205,\!954$	400,736	397,732	781,962	$61,\!399,\!733$
Percentage of househo	olds expos	ed to natural	hazards		
Earthquakes	0	100	100	$55(^{\star})$	59(*)
Volcanic eruptions	0	30	100	65	0
Wind effects	0	100	100	100	8(*)
Tsunamis or floods	$85(^{\dagger})$	84	100	100	$21(^{\dagger})$
Grounds movements	70	100	100	100	19
Forest fires	0	0	0	100	19
Avalanches	0	0	0	0	1
Percentage of households insured for their main home					
	52	43	50	59	99

Table 2: Population, exposure to major natural risks and insurance penetration rate for main home in France in 2006

Notes: ( $\star$ ): Réunion and continental France are exposed to earthquake of small intensity; continental France is also exposed to wind effects of small intensity. ( $\dagger$ ): the tsunamis to which French Guiana and continental France are exposed are also of small intensity, but these two areas are exposed to floods of high intensity. Population census by the French National Institute of Statistics and Economics Studies (INSEE) in 2006; GASPAR database by the French Ministry of Ecology; French Household Budget survey by INSEE in 2006 (13,374 observations for percentage calculations).

### 2.2 Insurance demand

Several reasons may explain a low demand for natural disasters insurance: perception biases, unaffordable insurance, uninsurable housing, anticipated assistance, which crowds out insurance, and a vicious circle of underinsurance.

**Insurance obligations.** Purchasing home insurance is often required as a condition for obtaining a mortgage. However, some homeowners with outstanding loans may not renew their insurance contracts once they have settled in. Indeed, as there is very little monitoring once people have moved in, some households choose to cancel insurance expenditure as soon as possible. This situation prevails in most Caribbean countries (Auffret, 2003).<sup>26</sup> In the French overseas departments, purchas-

 $<sup>^{26}</sup>$ This is also the case in the United States: banks or financial institutions can require the purchase of flood insurance to deliver a mortgage (Browne and Hoyt (2000), Office (1983)); there is

ing home insurance is also compulsory for tenants. According to my data, in 2006 only 67% tenants and 72% homeowners with outstanding loans are insured, whereas the overall insurance penetration rate is 48%.

**Perceptions biases.** Perception biases are often evoked to explain a low demand for coverage against extreme events. A wide literature deals with cognitive biases in the perception of extreme risks and their impact on demand for natural disasters insurance (see Tallon and Vergnaud (2007) for a review). For example, perception of low probabilities is reduced because of availability bias (Tversky and Kahneman, 1973), the "gambler's fallacy" following from a "belief in the law of small numbers" (Tversky and Kahneman, 1981),<sup>27</sup> the presence of a minimum threshold to look for information because of information cost (Kunreuther and Pauly, 2004) or the disability of comparing with ordinary risks (Kunreuther et al., 2001).

**Insurance affordability.** Another standard explanation is that insurance may be too expensive for overseas households. When insurance is available, insurance premiums offered to households are high in Latin America and the Caribbean (Auffret (2003), Charvériat (2000) and Evans (1996)). For example, in Mexico, premiums in earthquake-prone areas amount to 0.5% of the value of housing on an annual basis (Charvériat, 2000); in the Caribbean, rates for property insurance against disasters exceeded 1% of amount insured in the 1990s (Charvériat (2000), WB (1999)).

Even if insurance price is regulated in the French overseas departments, overseas households may not afford insurance, given that the median standard of living in the French overseas departments is almost 40% lower than in continental France (Michel et al., 2010).

very little monitoring of insurance renewal and many households do not renew their flood insurance policies (Kunreuther and Pauly, 2005).

<sup>&</sup>lt;sup>27</sup>The belief in the law of small numbers is the belief that, once the dwelling has been damaged by a disaster, the probability of being touched again is lower.

**Uninsurable housing.** In developing countries, many dwellings are so little resilient to natural events that they can be considered as uninsurable by insurers. Uninsurable housing has a great magnitude in Latin America and the Caribbean. In Mexico, about 50% of total housing stock correspond to uninsurable houses, built with no solid materials or without access to potable water (Charvériat, 2000). 60% of total housing stock in the Caribbean is built without any technical report (IDB, 2000).

In the French overseas departments, dwellings made of light materials (such as wood or sheet metal) of heterogenous quality represent 13% of dwellings in 2006 (Castéran and Ricroch, 2008) and are especially numerous in French Guiana. According to my data, the number of low-quality dwellings is significant and the insurance penetration rate is lower among their occupiers: this rate is 17% only among houses still in construction (which represent 3% of the sample), 15% among houses without hot water (23% of the sample), 34% among houses without drainage (53% of the sample) and 9% among houses without toilets inside the building (4% of the sample), whereas the overall insurance penetration rate is 48%.

All these low-quality dwellings are legal. A building permit, similar to the one existing in continental France or in other developed countries, is not required by law to build a house. Indeed, in the French overseas departments, the building property allows households to own the walls of their dwelling without owning the ground on which it is built. This is why more than 30% of individual dwellings in the French overseas departments have been built without a permit (DIREN (2005), Garnesson and Hecquet (2007), Olive and Riviere (2010)). Similarly, in the Caribbean region, building standards and location restrictions are either nonexistent or outdated and inadequate (Auffret, 2003). However, a building permit can be required by insurers as a condition for obtaining home insurance.

Uninsurable housing can be seen as a rational adaptation to exposure to natural

disasters in low-income countries and an illustration of the poverty trap. Low-income households use these either nonexistent or outdated and inadequate building rights to build low-quality dwellings, that would be destroyed by an impending natural disaster. These households with few assets can be trapped this way in chronic low-quality dwelling. This phenomenon has been studied, especially in the field of small businesses and agriculture (Barnett et al., 2008). Within the French overseas departments, however, according to my data dwellings of good quality are on average built in more exposed areas, probably because of risk exposure also provides positive amenities (river sight, fertile ground).

**Charity hazard.** Assistance is a substitute for formal insurance and decreases demand for insurance. This phenomenon, called charity hazard, is a typical example of Samaritan's dilemma. Charity hazard has been formalized by several theoretical papers but only few empirical findings have been established in the case of natural disasters insurance (see Raschky et al. (2010) for a review).<sup>28</sup>

Charity hazard has an important magnitude in many developing countries (Gilbert, 2001), including Latin America and the Caribbean. Indeed, the Caribbean region largely depends on international assistance: the World Bank and the Inter-American Development Bank provide important and increasing assistance to victims of natural disasters (Auffret, 2003).

In the French overseas departments, after natural disasters, households can also rely on substantial financial assistance by government, local authorities, non-governmental organizations or relatives. Their anticipation of financial assistance is essentially based on their past experience and is difficult to quantify because of the numerous assistance channels. Official statements after natural disasters confirm that the uninsured can rely on an important compensation from the government (Senate, 2005).

<sup>&</sup>lt;sup>28</sup>Charity hazard has been tested in health insurance (Herring (2005), Chernew et al. (2005), Brown and Finkelstein (2008)); first results suggest charity hazard in crop insurance in the United States (Deryugina and Kirwan, 2012).

Indeed, a main channel of governmental assistance to overseas France is the rescue fund for overseas. This compensation covers damages caused by natural disasters in the main home (including rebuilding); it is funded by budgetary credits.<sup>29</sup>

A vicious circle of uninsurance. Finally, underinsurance may reinforce itself for two main reasons. The first reason is similar to a peer effect. Social norms impact the decision to purchase insurance: individuals may decide to purchase insurance because they know others who did so; they may think that their relatives have similar preferences to them or have already spent the search costs of gathering information on risk, insurance or relief (Kunreuther and Pauly, 2005).

The second reason relies on the endogeneity of provided assistance and is this way linked to charity hazard. The neighbors' decision of uninsurance increases the neighborhood eligibility for assistance and so decreases the individual benefit of purchasing insurance. In other words, the more people are uninsured around one individual, the less he needs to purchase insurance since the political power of the uninsured grows. This mechanism is predicted by theory for many types of public aid: Arvan and Nickerson (2006) consider endogenous governmental compensation and show that an individual's purchase of insurance coverage creates negative externalities by diminishing neighborhood eligibility for such aid.<sup>30</sup>

# 3 Theoretical model

I estimate a semi-structural model of insurance supply and demand (Abel (1986), Pauly (1974), Rothschild and Stiglitz (1976)) within the French overseas departments. I detail here the theoretical specification of this model.

The supply equation explains the insurance premium offered by insurers. This price

 $<sup>^{29}\</sup>mathrm{See}$  order of December 8, 2010 relative to the implementation of help facility by the rescue fund for overseas.

 $<sup>^{30}</sup>$ See Herring (2005) for illustration of endogenous availability of charity care for health.

conveys the nullity of the insurers' expected profit. Nullity of expected profit means that collected premiums equal expected losses. Both amounts, which reflect the specific design of the French natural disaster insurance system (Section 2), are here precisely modeled. Insurers are assumed to offer a unique standardized contract with full coverage.

The demand equation explains the household's probability of purchasing insurance. The quantity of purchased insurance results from the comparison by households between their expected utilities with and without insurance. The decision of whether to purchase insurance or not depends on the insurance price. I enrich this demand equation to precisely model the underlying determinants of insurance demand (Section 2).

### 3.1 Risk structure

A dwelling suffers a loss  $L_d$  caused by natural disasters with probability  $p_d$ . I assume that uninsured households receive assistance  $A_d$  after a disaster. The net loss is thus  $L_d - A_d$ . Ordinary risks (such as theft, fire, explosion or water damages) cause a loss  $L_o$  with probability  $p_o$ . No assistance is provided to compensate damages caused by these individual risks.

For the sake of simplicity, losses caused by natural disasters and damages caused by ordinary risks are assumed to be independent events. As the product of the two probabilities  $p_d p_o$  is negligible with respect to any of the two probabilities, there are indeed three states of Nature: a high loss  $L_d - A_d$  with a small probability  $p_d$ , a small loss  $L_o$  with an important probability  $p_o$ , and no loss with probability  $1 - p_d - p_o$ (Figure 2).

Households may have a potentially biased risk assessment, different from the true ones (that are insurers' ones), for the probability of ordinary losses  $\tilde{p}_o$ , for the prob-

$$\frac{p_d p_o}{p_d (1 - p_o) \approx p_d} W - L_d + A_d$$

$$\frac{(1 - p_d) p_o \approx p_o}{(1 - p_d) (1 - p_o)} W - L_o$$

$$\frac{(1 - p_d) (1 - p_o)}{\approx 1 - p_d - p_o} W$$

Figure 2: Risk structure

ability of natural disasters  $\tilde{p}_d$  and for the losses  $\tilde{L}_d$  caused by natural disasters. For the sake of simplicity, I assume that households have the same estimation of their ordinary losses  $L_o$  than insurers.

### 3.2 Modeling the supply side

**Insurance policy.** As households' choices of coverage are restrained to furniture in France (Section 2), I assume that a unique standardized contract with full coverage is offered by insurers. Therefore households either purchase home insurance ( $\alpha = 1$ ) or not ( $\alpha = 0$ ).

Nullity of expected profit. Insurance companies are assumed to be price takers. Competition on insurance market and risk neutrality of insurers imply the nullity of the insurers' expected profit on each group of identical households (for what is observed by the insurers). Nullity of expected profit means that collected premiums equal expected losses caused by ordinary risks  $EL_o$  and by natural disasters  $EL_d$ . I add a multiplicative constant c; this loading factor represents transaction costs (information search, negotiation, policy drafting, controls, claim disputes).

$$\pi = c(EL_o + EL_d). \tag{1}$$

Collected premiums and expected losses both reflect the specific design of the French natural disaster insurance system; they are here precisely modeled.

**Premiums.** The home insurance premium  $\pi$  is the sum of the premium for natural disasters  $\pi_d$  and the premium for other risks  $\pi_o$ . The premium for natural disasters  $\pi_d$  amounts to r = 0.12 of the premium for other risks  $\pi_o$  (Section 2).

$$\left. \begin{array}{l} \pi = \pi_d + \pi_o, \\ \pi_d = r\pi_o. \end{array} \right\} \Rightarrow \pi = \frac{1+r}{r}\pi_d.$$

$$(2)$$

Expected losses. Expected ordinary losses equal

$$EL_o = p_o L_o. (3)$$

All insurers are assumed to be reinsured against natural disasters by CCR, since CCR captures more than 90% of the market shares on the natural disasters reinsurance market (Section 2). Expected losses caused by natural disasters are determined by the unactuarial reinsurance policy offered by CCR (Section 2). The insurer is exposed until a deductible, which is the natural disaster premium. In counterpart, he has to pay a fixed share k = 0.635 of the natural disaster premium. Indeed, this share corresponds to the sum of the price of reinsurance policy and a tax to fund prevention measures (Section 2).

$$EL_d = p_d \min\left(\pi_d, \frac{L_d}{2}\right) + k\pi_d,\tag{4}$$

$$= (p_d + k)\pi_d. \tag{5}$$

as  $\pi_d < \frac{L_d}{2}$ .

The supply equation (1) becomes

$$\pi = c(EL_o + EL_d),\tag{6}$$

$$= c(p_o L_o + (p_d + k)\pi_d),$$
(7)

$$= cp_o L_o + c(p_d + k) \frac{r}{1+r} \pi.$$
 (8)

Thus

$$\log(\pi) = \log(cp_oL_o) - \log\left(1 - ck\frac{r}{1+r} - cp_d\frac{r}{1+r}\right).$$
(9)

### 3.3 Modeling the demand side

**Comparison of expected utilities.** A household is assumed to be risk averse: his utility function  $U(\cdot)$  is concave with respect to his wealth. He purchases insurance  $(\alpha = 1)$  if and only if his expected utility EU is higher when he is insured  $(\alpha = 1)$ than when he is not  $(\alpha = 0)$ .<sup>31</sup>

$$\alpha = 1 \Leftrightarrow EU|_{\alpha=1} \ge EU|_{\alpha=0}.$$
(10)

Given full insurance of price  $\pi$ , the expected utility of the insured is

$$EU|_{\alpha=1} = U(W - \pi).$$
 (11)

The expected utility of the uninsured is

$$EU|_{\alpha=0} = \tilde{p}_o U(W - L_o) + \tilde{p}_d U(W - \tilde{L}_d + \tilde{A}_d) + (1 - \tilde{p}_o - \tilde{p}_d)U(W),$$
  
=  $U(W) - \tilde{p}_o[U(W) - U(W - L_o)] - \tilde{p}_d[U(W) - U(W - \tilde{L}_d + \tilde{A}_d)].$  (12)

I enrich the demand equation (10) to precisely model the underlying determinants of insurance demand (Section 2).

 $<sup>^{31}</sup>$ The standard expected utility framework may not be the best appropriate to analyze the economic consequences of fat-tailed events (Weitzman, 2009); however Weitzman's alternatives may be less appropriate to study the purchase of catastrophe insurance.

**Insurance obligations.** Purchasing home insurance is compulsory for tenants and often required as a condition for obtaining a mortgage (Section 2). As many tenants and homeowners with outstanding loans remain uninsured (Section 2), proxies  $\{O_k\}_k$  for occupancy status are added to control for these insurance obligations and to measure their impact.

Uninsurable housing. A significant number of houses are uninsurable buildings: they do not meet building standards or have been realized without building permit (Section 2). I enrich the demand equation (10) by adding proxies  $\{H_k\}_k$  for uninsurable housing.

**Peer effects.** To test whether the more neighbors are insured, the higher the individual probability of purchasing insurance is, I add in the demand equation the expected penetration rate  $E(Z_{\text{peer},i})$  of the group  $J_{\text{peer}}$  of peers to which the household *i* belongs:

$$E(Z_{\text{peer},i}) = \frac{\sum_{j \in J_{\text{peer}}, j \neq i} \alpha(j)}{\operatorname{card}(J_{\text{peer}}) - 1}.$$
(13)

This model corresponds to a degenerated Nash equilibrium, where the decision of the group impacts the household's decision but where the reverse impact is negligible because of the size of each group. This strategy is inspired by other papers studying peer effects, such as Hernández-Murillo and Sengupta (2012).

Neighborhood eligibility for assistance. The decision of purchasing insurance by one household depends on the decisions of others', not only via peer effects but via neighborhood eligibility for assistance. To test for its endogenous nature, anticipated assistance is assumed to depend on the expected penetration rate  $E(Z_{aid})$  of the group  $J_{aid}$  for aid eligibility:  $\tilde{A}_d(E(Z_{aid}))$ .<sup>32</sup> This enables to test whether the percentage of insured households around one individual decreases his likelihood to get assistance after a disaster and so decreases the charity hazard effect.

<sup>&</sup>lt;sup>32</sup>Again, this model corresponds to a degenerated Nash equilibrium.

The demand equation becomes

$$\alpha = 1 \Leftrightarrow [U(W - \pi) - U(W)] + \tilde{p}_o[U(W) - U(W - L_o)] + \tilde{p}_d[U(W) - U(W - \tilde{L}_d + \tilde{A}_d(E(Z_{aid})))] + \sum_k o_k O_k + \sum_k h_k H_k + \delta E(Z_{peer}) \ge 0.$$
(14)

# 4 Data and model specification, identification and calibration

I present the unique household-level micro-database that has been built to estimate this theoretical model (Section 3). The empirical specification, which is fully parametric, is detailed. The identification and calibration of the model are discussed; all the performed robustness tests are exposed.

### 4.1 Data

The database combines information about insurance expenditure for the insured, risk exposure and other economic variables for the insured and the uninsured. It has been built by matching the 2006 French Household Budget survey with the GASPAR database, which presents information about exposure to natural disasters.

The French Household Budget survey, managed by the French National Institute of Statistics and Economics Studies (INSEE), is a comprehensive national survey on households' expenditures, and in particular insurance expenditures. Regarding home insurance, households declare whether they have purchased home insurance and if so the amount of the premium paid. Neither the identity of the different insurers nor the type of companies (mutual insurance company or not) are observed. Data on assistance are not available (Section 4).<sup>33</sup> The French Household Budget survey also provides information about the household himself (such as size, income and standard of living, and, for the reference person,<sup>34</sup> gender, age and place of birth). Detailed information about housing (such as occupancy status, dwelling quality, number of rooms) is given. However, no information on dwelling compliance with building standards and permits is provided. The 2006 French Household Budget survey comprises 3,134 households living in the French overseas departments.<sup>35</sup>

The GASPAR database, built by the French Ministry of Ecology, is the database of computer-aided management for administrative procedures relative to natural and technological risks. It specifies which hazard each municipality is exposed to, these five hazards being earthquakes, volcanic eruptions, hurricanes or cyclones, tsunamis or floods, and grounds movements.<sup>36</sup> It also provides the number of disasters by hazard type in each municipality from 1990 (date of the enforcement of the natural disasters insurance system in the French overseas departments) to the survey date.

As the decision of whether to purchase insurance or not depends on the insurance price, I exclude from the study the households insured by their relatives or their employer and all the other households who declare themselves as insured but do not report their premium amount. Over the initial 3,134 households, 2,860 observations remain. I then exclude 40 observations for which key variables (annual income, number of rooms) are missing and 11 for which the declared annual income is below  $\notin$ 500. Finally, 2,809 observations remain.

<sup>&</sup>lt;sup>33</sup>Even detailed data on assistance provided by the rescue fund for overseas are not available. Annual aggregate statistics at the departmental level only were provided by the French Ministry of Overseas.

<sup>&</sup>lt;sup>34</sup>More often than not, the household reference person is the family reference person when there is one, or the oldest man, with priority to the oldest active person.

<sup>&</sup>lt;sup>35</sup>In French Guiana, the sampling plan of the 2006 French Household Budget survey overrepresents the coastal area, which is more exposed to floods and tsunamis (Forgeot and Celma, 2009).

 $<sup>^{36}</sup>$ It also specifies whether each municipality is exposed to forest fires, but this hazard is considered as insurable and therefore it is not considered as a natural disaster (Section 2). See Table 2 for the exposure of each French overseas department to these different natural hazards.

Table 3 describes my sample. The average municipal exposure to natural risks is high but very heterogenous: while municipalities are on average exposed to 4 distinct natural hazards, some are exposed to 5 hazards according to the GASPAR database, and others to none; on average, 8 natural disasters have occurred since 1990, this number reaching 18 in some municipalities, whereas others have been spared. 48% of households living in the French overseas departments have purchased home insurance, which includes the coverage against natural disasters, for their main home in 2006. This insurance rate also strongly varies among municipalities: it reaches 0.92% in some of them, whereas in others no one is insured. The premium paid by the insured is on average  $\notin 254$  and ranges from  $\notin 20$  to  $\notin 2,000$ , reflecting important disparities among the sample population. Indeed, the annual income ranges from  $\notin 600$  to  $\notin 169,637$  for an average of  $\notin 22,694$ . 36% of households are tenants; 13% are homeowners with outstanding loans, the remainder owning their home. While dwellings present 4 rooms on average, some present only 1, others 12. Many houses lack some modern conveniences: 23% are without hot water; 53% without drainage; 4% without toilets inside the building. 3% of houses are still in construction. Finally, the reference person is born in continental France and abroad in 10% and 8% of cases, respectively; she is a woman in 46% of households; her age varies between 17 and 95.

# 4.2 Specification, identification and calibration of the supply side

As no information is provided on the insurer, the nullity of the expected profit over all insurers confounded only can be considered.<sup>37</sup>

<sup>&</sup>lt;sup>37</sup>However, some characteristics of policy-holders could capture their choice of insurance company (and this way indirectly modify the premium, even if insurers may not measure them). For example, some insurance companies (mainly mutual ones) cover civil servants exclusively and are said to increase premiums with respect to the risk exposure to a smaller extent than other companies. Here dummies for civil servants and other characteristics such as age or gender appear as non significant in the premium estimation.

	Percentage / mean	Minimum	Maximum
Number of natural hazards	4	0	5
Number of past natural disasters	8	0	18
Households insured for their main home	48%		
Insured households living in the same municipality	47%	0%	0.92%
Premium paid by the insured	€254	€20	€2,000
Annual income	€22,694	€600	€169,637
Standard of living	€13,359	€407	€87,266
Number of rooms	4	1	12
Tenants	36%		
Homebuyers	13%		
Houses still in construction	3%		
Houses without hot water	23%		
Houses without drainage	53%		
Houses without toilets inside the building	4%		
Reference person born in continental France	10%		
Reference person born abroad	8%		
Age of the reference person	49	17	95
Gender (woman) of the reference person	46%		

### Table 3: Descriptive statistics

*Notes:* Are considered in the sum of natural hazards earthquakes, volcanic eruptions, wind effects, floods (including tsunamis) and ground movements. 2006 French Household Budget survey and GASPAR database. 2,809 observations.

**Ordinary losses.** Ordinary losses  $L_o$  depend on the dwelling characteristics. More precisely, they depend on the values of furniture and of the building. One proxy for the value of furniture (mainly jewels and furniture) is the standard of living Y, i.e. the income divided by the household's size.<sup>38</sup> One proxy for the value of the building is the number of rooms N. Losses also depend on occupancy status, since tenants, denoted by  $O_t = 1$ , do not bear all losses, a part of them being borne by their landlord.<sup>39</sup>

These effects are assumed to be multiplicative: the value of furniture in each room

 $<sup>^{38}</sup>$ The standard of living is measured by the income per consumption unit. The first adult is worth one consumption unit; the second adult and each child older than 14 are worth 0.5; younger children are worth 0.3.

<sup>&</sup>lt;sup>39</sup>Landlord is responsible for potential damages to furniture in furnished dwellings, to the structure (walls, foundations) and for damages implying his liability (structural defects).

increases with respect to the standard of living Y, and the number of pieces of furniture increases with respect to the number of rooms N; last, tenants insure only a fraction  $(1 - \tau)$ ,  $\tau \ge 0$  of the total value of the dwelling. l is a multiplicative constant. Thus, the ordinary loss  $L_{oi}$  for household i is

$$L_{oi} = lY_i^y N_i^n (1 - \tau O_{ti}), \ \tau \ge 0.$$
(15)

y and n are the elasticities of the loss with respect to the standard of living and the number of rooms, respectively.<sup>40</sup>

Loss probabilities. I have no specific information on  $p_o$ , since I do not observe past ordinary losses nor other proxies for the probability of suffering these losses.

Insurers estimate the probability of natural disasters using information about physical hazards. Business practices indicate that French insurers use very basic information about natural risk exposure, very likely because their financial exposure to natural risk is limited thanks to the reinsurance contract offered by CCR (Section 2); this is a typical case of *ex ante* moral hazard. I assume that the probability of natural disaster estimated by insurers for each household *i* linearly increases with respect to the sum of hazards  $R_i$  to which his municipality is exposed.<sup>41</sup>

Insurers: 
$$p_{di} = pR_i, \ p \ge 0.$$
 (16)

**Error.** An error  $\epsilon$  is attached to the supply equation. This error term is due to a potential assessment error made by the insurer. It is assumed to be normally distributed.

<sup>&</sup>lt;sup>40</sup>Losses caused by natural disasters are not estimated in the supply equation, as potential loss and reinsurance premium paid by the insurer are not determined by these losses (Section 2).

 $<sup>^{41}</sup>R$  is public information.

Using (9), (15) and (16), I get

$$\log(\pi_i) = \log(cp_o l) + y \log(Y_i) + n \log(N_i) + \log(1 - \tau O_{ti}) - \log\left(1 - \frac{ckr}{1 + r} - \frac{cpr}{1 + r}R_i\right) + \sigma \epsilon$$
$$= c_{\pi} + y \log(Y_i) + n \log(N_i) + \log(1 - \tau O_{ti}) - \log(1 - \kappa - \rho R_i) + \sigma \epsilon, \quad (17)$$

where  $c_{\pi} = \log(cp_o l)$ ,  $\kappa = ckr/(1+r)$  and  $\rho = cpr/(1+r)$ .

Identification and calibration of risk parameters.  $c_{\pi}$ ,  $1 - \kappa$  and  $\rho$  cannot be simultaneously identified. I estimate  $c_{\pi}$  and  $\rho$  and I calibrate  $\kappa = ckr/(1+r)$ . r = 0.12 and k = 0.635 are imposed by the government and CCR (Section 2). I calibrate loading factor c using values provided by literature:  $c \approx 1.3$  (Gollier, 2003). Thus, I take  $\kappa = ckr/(1+r) \approx 0.088$ . Estimations are performed for  $c \in \{1, 1.5\}$ , that is for  $\kappa \in \{0.068, 0.10\}$ .

Estimation of  $c_{\pi} = \log(cp_o l)$  does not enable to simultaneously identify  $p_o$  and l (and c), even when considering that c is already calibrated. I calibrate  $p_o$  using statistics provided in continental France:  $p_o \approx 0.075$  (FFSA, 2006);<sup>42</sup> estimations are performed for  $p_o \in (0.05, 0.5)$ . Risk parameter l is deduced from the estimated value of  $c_{\pi}$ .

Similarly, the risk parameter p will derive from the estimated value of  $\rho = cpr/(1+r)$ , given that c is calibrated and r is known.

Given that  $\alpha_i$  states whether the household *i* purchases insurance or not, the supply

<sup>&</sup>lt;sup>42</sup>In continental France, between 2000 and 2004, statistics by the French federation of insurance companies about home insurance show that the frequency of ordinary risks is around  $p_o \approx 0.075$ (FFSA, 2006). Abroad, the probabilities of some of the ordinary risks correspond to the same order of magnitude. In Taiwan, the probability of fire occurrences in residential buildings per m<sup>2</sup> of floor space is around 0.01 (Lin, 2005). In Long Beach (CA), the probability of burglary is around 1.9% for a house which has never been burglarized and reaches 59% after a first burglary (Short et al., 2009).

equation becomes

$$\begin{cases} \text{if } \alpha_{i} = 1, \ \log(\pi_{i}) = c_{\pi} + y \log(Y_{i}) + n \log(N_{i}) + \log(1 - \tau O_{ti}) \\ -\log(1 - \kappa - \rho R_{i}) + \sigma \epsilon_{i}, \\ \text{if } \alpha_{i} = 0, \ \pi_{i} = 0, \end{cases}$$

where  $c_{\pi} = \log(cp_o l)$  and  $\rho = cpr/(1+r)$  are estimated parameters and  $\kappa = ckr/(1+r)$  is calibrated.

# 4.3 Specification, identification and calibration of the demand side

Utility function and risk aversion. In an expected utility setting, constant relative risk aversion is a reasonably good approximation of individual attitude toward risk (Chiappori and Salanié, 2008). A constant relative risk aversion  $\lambda$  with respect to the income corresponds to the following utility function:  $U(W) = W^{1-\lambda}/(1-\lambda)$ . Literature has estimated different values for  $\lambda$  (Chiappori and Salanié, 2008). Estimations are here performed under the assumptions that utility is the log function, which is the limit case of  $U(W) = W^{1-\lambda}/(1-\lambda)$  as  $\lambda$  tends to 1. Results are robust when using  $\lambda = 2$  or  $\lambda = 3$ .<sup>43</sup>

Losses and loss probabilities. As households are assumed to have the same estimation as insurers of their ordinary losses  $L_o$  (Section 3), losses  $L_o$  are simultaneously estimated in the supply equation - but for the insured households only and in the demand equation. On the contrary, losses  $\tilde{L}_d$  intervene in the demand

<sup>&</sup>lt;sup>43</sup>Estimation of risk aversion raises numerical problems. Indeed, risk aversion determines the orders of magnitude of the terms expressing the expected utility losses; and if the orders of magnitude of the variables in the demand equation strongly differ, the model may be misestimated (coefficients corresponding to the negligible terms may appear as non significant). For example, in the case of the log function, I use  $U(W) = c_U \log(W)$ , with  $c_U = 10$ . Indeed, with  $c_U = 1$  the terms expressing the expected utility losses would be too small by comparison with the other terms (Equation 14) and the corresponding coefficients would be poorly estimated. The adequate value of  $c_U$  would be different when using another value of risk aversion  $\lambda$ . Note that  $c_U$  and  $\lambda$  cannot be simultaneously identified.

equation only.<sup>44</sup> Losses  $\tilde{L}_d$  caused by natural disasters fundamentally depend on the same dwelling characteristics than ordinary losses  $L_o$ . For the sake of simplicity, I assume that, for every household i,

$$\tilde{L}_{di} = \beta L_{oi}, \beta \ge 1. \tag{18}$$

Because of this intrinsic link between ordinary losses and losses caused by natural disasters (that remains even in a nonproportional specification), the utility decrease caused by ordinary losses, weighted by their occurrence probability,  $\tilde{p}_o[U(W) - U(W - L_o)]$ , and the utility decrease caused by natural disasters, weighted by their occurrence probability,  $\tilde{p}_d[U(W) - U(W - L_d + \tilde{A}_d)]$ , are fundamentally linked and  $(\tilde{p}_o, \tilde{p}_d, \beta)$  cannot be simultaneously identified. I favor the estimation of the natural disasters parameters, which enables to capture charity hazard, and I calibrate  $\beta$  and  $\tilde{p}_o$ .<sup>45</sup>

In continental France, the ratio of the mean natural disasters losses over the mean ordinary losses  $\bar{L}_d/\bar{L}_o$  ranges from 6.25 to 12.5.<sup>46</sup> Knowing that natural disasters are more intense events in the French overseas departments, I take  $\beta = 15$ . As a sensitivity test, I have performed estimations for  $\beta \in (10, 20)$  and significative-ness and sign of all estimated coefficients are robust to the choice of this parameter.<sup>47</sup>

The probability of ordinary losses  $\tilde{p}_o$ , of which no proxies is observed. Section 5 presents the results under the assumptions that  $\tilde{p}_o = 0.075$ . Estimations are per-

<sup>&</sup>lt;sup>44</sup>Losses caused by natural disasters are not estimated in the supply equation, as potential loss and reinsurance premium paid by the insurer are not determined by these losses (Section 2).

<sup>&</sup>lt;sup>45</sup>Even once  $\beta$  is calibrated,  $\tilde{p}_o$  and a fixed part a in an affine function  $\tilde{p}_d(S) = a + bS$  are not simultaneously identified.

<sup>&</sup>lt;sup>46</sup>In continental France, between 2000 and 2004, statistics by the French federation of insurers companies about home insurance show that the damages caused by ordinary risks are on average around  $\leq 1,200$  (FFSA, 2006). Damages caused by floods and ground movements for continental households are on average around  $\leq 7,500$  and  $\leq 15,000$ , respectively (Grislain-Letrémy and Peinturier, 2010).

<sup>&</sup>lt;sup>47</sup>Indeed, as the potential losses cannot exceed the wealth of the household, wealth determines the upper bound of the range of values for  $\beta$ . For  $\beta = 20$ , the potential losses already exceed the wealth of 17 households. Estimations provide consistent orders of magnitudes: losses  $L_o$  are between  $\leq 300$  and  $\leq 2,700$  (for  $\beta = 15$ ).

formed for  $\tilde{p}_o \in (0.05, 0.5)$ , while allowing  $\tilde{p}_o$  to be different from  $p_o$ . Significativeness and sign of all estimated coefficients are robust to the choice of these parameters.

Learning from past disasters. The number S of past disasters that have occurred in the municipality between the enforcement of the insurance system (1990) to the sampling date (2006) is public information. Past disasters have a double impact on households' estimation of their exposure to natural disasters. First, the number of past disasters increases households' estimation of their probability  $\tilde{p}_d$  of suffering another disaster. Second, this number modifies households' expectation of receiving assistance, since their expectation is built on compensation provided to them after past events. Thus, households' expected assistance  $\tilde{A}_d$  depends on number S of past disasters and on penetration rate  $E(Z_{aid})$  of the group  $J_a$  of joint eligibility for assistance (Section 3):  $\tilde{A}_d(S, E(Z_{aid}))$ . Given that no proxy for expected assistance is observed (Section 4), capturing charity hazard requires to disentangle the two impacts of the number S of past disasters on insurance demand.

More formally, in the theoretical model,

$$\alpha = 1 \Leftrightarrow \left[ \log(W - \pi) - \log(W) \right] + \tilde{p}_o \left[ \log(W) - \log(W - L_o) \right]$$
  
+  $q_d(S, E(Z_{aid})) \left[ \log(W) - \log(W - \beta L_o) \right] + \sum_k o_k O_k$   
+  $\sum_k h_k H_k + \delta E(Z_{peer}) + \nu \epsilon + \eta \ge 0,$  (19)

where  $q_d(S, E(Z_{aid}))$  "summarizes" the two impacts of past disasters, the one of the probability  $\tilde{p}_d$  of natural disasters and the one on expected assistance  $\tilde{A}_d$ .<sup>48</sup> Indeed, as the number S of past disasters increases, insurance demand is modified by a premium increase and a utility loss. The premium increase effect (PIE) refers to the fact that an increase of insurance price (as risk and sinistrality are correlated) may reduce insurance demand. The utility loss effect (ULE) denotes the fact that the

<sup>&</sup>lt;sup>48</sup>Simultaneous estimation of functional forms of  $\tilde{p}_d$  and  $A_d$  with respect to S leads to non robust results, as it can lead either to positive values, as expected, or to a huge amount of assistance (beyond loss, i.e.  $A_d > L_d$ ) that would make the utility decrease positive: households would gain in the case of natural disasters and then  $\tilde{p}_d$  becomes negative to balance this effect.

anticipated loss of utility may also increase, which should on the contrary increase insurance demand. If anticipated assistance also increases with respect to the number of past disasters, this reduces the utility loss effect: this is the charity hazard effect (CHE). The sign of  $\partial \alpha / \partial S$  is determined by the sign of

$$\underbrace{-\frac{\partial \pi}{\partial p_{d}}\frac{\partial p_{d}}{\partial S}\frac{dU}{dW}}_{\text{PIE}\leq0} \underbrace{-\frac{\partial \tilde{p}_{d}}{\partial S}(U(W-\tilde{L}_{d}+\tilde{A}_{d}(S))-U(W))}_{\text{ULE}\geq0} \underbrace{-\frac{\tilde{p}_{d}(S)}{\geq0}\underbrace{\frac{\partial \tilde{A}_{d}}{\partial S}}_{\geq0}\underbrace{U'(W-\tilde{L}_{d}+\tilde{A}_{d}(S))}_{\geq0}}_{\text{CHE}\leq0}_{\text{ULE}+\text{CHE}=-\frac{\partial q_{d}}{\partial S}\text{UL}}$$
(20)

As data on anticipated assistance are not available, the sum of ULE and CHE only can be identified. Thus, estimation reveals the existence of a charity hazard effect only if it exceeds the utility loss effect, i.e. only if  $|CHE| \ge ULE$  that is only if  $\frac{\partial q_d}{\partial S} \le 0$ . If on the contrary  $\frac{\partial q_d}{\partial S} > 0$ , |CHE| < ULE and this would be consistent with a small or null value of the charity hazard effect, CHE.

$$\frac{\partial q_d}{\partial S} \le 0 \Leftrightarrow |\text{CHE}| \ge \text{ULE} \Rightarrow \frac{\partial \tilde{A}_d}{\partial S} \ge 0.$$
(21)

Indeed, a negative sign of  $\partial q_d/\partial S$  would indicate that anticipated assistance by households increases with respect to the number S of past disasters that have occurred in the municipality (Equation 20). This would correspond to a cumulative effect in the anticipation of assistance: households living in municipalities where numerous disasters occurred have noticed the importance of assistance, probably more than the other households have; therefore they anticipate higher amounts of *ex post* aid.

Besides, to test for the endogenous nature of anticipated assistance (Section 3), the expected penetration rate  $E(Z_{aid})$  of the group  $J_{aid}$  of joint eligibility for assistance is crossed with the charity hazard effect. Thus, for each household i,

$$q_{di} = (q + \theta E(Z_{\text{aid},i}))S_i.$$
(22)

A negative sign of q would indicate a charity hazard effect and a positive sign of  $\theta$  would mean that the percentage of insured neighbors decreases this charity hazard effect, as it decreases the individual likelihood to get assistance after a disaster.

Let us check that no other phenomenon could imply a negative sign of q. First, perception bias could decrease  $\partial \tilde{p}_d / \partial S$  and therefore the demand for insurance (Section 2) but not imply a negative sign for the estimated coefficient  $\partial q_d / \partial S$ : even in the presence of perception bias, the perceived utility loss would still increases with respect to the number S of past disasters, even when considering the belief in the law of small numbers. This belief consists in considering that once a dwelling has been damaged by a disaster, the probability of being struck again is lower (Tversky and Kahneman, 1981). Households may have this belief after one unique disaster, but likely not after having been struck several times, which is the case as they have suffered in average 8 disasters since 1990 (Table 3). Second, a negative sign of q could derive from uncontrolled differences in risk aversion. In other words, one cannot exclude for now that more exposed households do not purchase insurance because they have a lower risk aversion.<sup>49</sup> However, data show that households who live in more exposed areas are presumed to be actually more risk averse (Table 4): they are older, the proportion of women among them is higher.<sup>50</sup> Besides, among households who live in more exposed areas, the proportions of people either born in continental land (who could be used to managing risk differently), or who purchase automobile insurance - with a limited or an extended coverage  $-^{51}$  are not significantly higher.<sup>52</sup>

<sup>&</sup>lt;sup>49</sup>Heterogeneities in wealth or in dwelling quality are already taken into account in the demand equation. The location choice of wealthy households is not significantly correlated with risk exposure; dwellings of good quality are on average built in more exposed areas (Section 2).

<sup>&</sup>lt;sup>50</sup>Levin et al. (1988), Powell and Ansic (1997), Halek and Eisenhauer (2001) and Jianakoplos and Bernasek (1998) show that women are more risk averse than men. Morin and Suarez (1983), Palsson (1996) show that the risk aversion increases with respect to the age; however, cohort effects may complicate the impact of age on risk aversion (Brown (1990), Jianakoplos and Bernasek (1998)). Besides, risk aversion depends on contextual framework (Schubert et al. (1999)).

 $<sup>^{51}{\</sup>rm Third}\text{-party}$  insurance only is mandatory for automobiles. Only 1.5% of households own a car without this coverage.

<sup>&</sup>lt;sup>52</sup>These statistics also confirm that supply accessibility is not lower for exposed households. Indeed, difficulties in terms of supply accessibility are especially limited for home insurance, as households can purchase a home insurance policy on the phone in approximately 20 minutes.

Table 4: Self-selection on housing market: correlation between proxies for risk aversion and the number of past disasters in the municipality

	Correlation	
	value	$\Pr >  \mathbf{r} $
Age of the reference person	0.060	0.0015
Gender (woman) of the reference person	0.068	0.0003
Place of birth (continental France) of the reference person	-0.0032	0.86
Insured automobile	-0.0053	0.78
Comprehensive automobile coverage	0.029	0.13

*Notes:* 2006 French Household Budget survey and GASPAR database. 2,809 observations.

Adverse selection because of insurance pricing? On the contrary, if  $q \ge 0$ ,

it can be tested whether there is adverse selection, that is whether

$$ULE + CHE \ge |PIE|, \tag{23}$$

i.e. whether  $\partial \alpha / \partial S \geq 0$ . Insurance subsidization for exposed households by the least exposed ones could lead to an extensive adverse selection, that is to a higher participation, on the insurance market, of exposed households. Here, adverse selection would mainly come not from the lack of information by insurers but from their limited incentive to use information because of reinsurance policies (Section 2): as insurers bear a very limited part of losses caused by natural disasters, insurance premiums only partially reflect natural risks and this way subsidies exposed households.

Insurance obligations. Dummies for tenants  $O_t$  and for homeowners with outstanding loans  $O_l$  are added to control for these insurance obligations and to measure their impact.<sup>53</sup> Results are robust when tenants and homeowners with outstanding loans are excluded from the sample and also when the model is estimated either on tenants only.<sup>54</sup>

<sup>&</sup>lt;sup>53</sup>Monitoring of insurance renewal may be partly realized in public housing. Unfortunately, information about public housing is not available.

 $<sup>^{54}\</sup>mathrm{An}$  estimation on homeowners with outstanding loans only is not possible, as they are 336 only.

Uninsurable housing. Data provide information about dwelling quality, but not about dwelling compliance with building standards or permits (Section 4). The Inter-American Development Bank defines insurable housing market as housing built by solid materials and with potable water and drainage (IDB, 2000). Here I control for uninsurability by adding dummies for low dwelling quality: a dummy  $H_c$  for houses still in construction and three dummies for houses without modern conveniences (without hot water  $H_w$ , without drainage  $H_d$  and without toilets inside the house  $H_t$ ).

Groups of peers and of joint eligibility for assistance. Different definitions for the group  $J_{\text{peer}}$  of peers and for the group  $J_{\text{aid}}$  for joint eligibility have been tested by crossing the municipal level (which is the smallest geographical level that I observe) with any other observed household characteristic (such as age, gender, occupational groups, place of birth).<sup>55</sup>

The place of birth can also explain the probability of purchasing insurance via an "initial peer effect". Indeed, as insurance penetration rate of continental France is exceptionally high (Grislain-Letrémy and Peinturier, 2010), having grown up in a place where a wide majority of people are insured can increase the probability of purchasing insurance. This is why dummies  $B_{cl}$  and  $B_a$  for households born in continental France and abroad, respectively are added to the demand equation.

Wealth. The wealth used to perform estimations corresponds to households' holdings. Indeed, households can lose almost all their possessions in the case of a natural disaster. For the sake of simplicity, I assume that the observed income w earned by the household during the year is constant over time until the death of the reference person inside the household. I denote by A the age of the reference person in the household and E his life expectancy, which is calculated by linear interpolation using registry office statistics (Niel and Beaumel, 2010). I use here the discount rates

<sup>&</sup>lt;sup>55</sup>When the geographical level is the municipality, the geographical impact is implicitly assumed to be uniform across municipalities, as there are too many municipalities to allow for different coefficients between municipalities.

recommended by Gollier (2007), that is  $r_1 = 4\%$  until 30 years and  $r_2 = 2\%$  beyond. Thus, I get

$$W = \sum_{0 \le t \le E-A} \frac{w}{(1+r_1)^t} \qquad \text{if } E - A \le 30, \qquad (24)$$

$$= \sum_{0 \le t \le 30} \frac{w}{(1+r_1)^t} + \sum_{31 \le t \le E-A} \frac{w}{(1+r_1)^{31}(1+r_2)^{t-31}} \text{ otherwise.}$$
(25)

This corresponds to a multiplication of the annual income by a factor that depends on the age of the reference person: it varies from 6 for the 95-year-old people to 24 for the 17-year-old ones, with an average of 18. Sign and significativeness of all coefficients are robust to this modification: they are similar when using the holdings as here defined or the annual income. They are even robust when uniformly multiplying annual income until to  $100.5^{6}$ 

Selection bias. I add the term  $\nu\epsilon$ , where  $\epsilon$  is the error attached to the insurance premium. This term allows for a selection bias, i.e. for correlation between unobserved heterogeneity factors that affect the insurance premium and the decision to purchase insurance.

**Error.** Another error  $\eta$  is also attached to the decision to purchase insurance. It can be interpreted as an assessment error made by households. It is also assumed to be normally distributed.  $\epsilon$  and  $\eta$  are assumed to be independent, since possible correlation is taken into account by the selection bias term.

<sup>&</sup>lt;sup>56</sup>Indeed, at the first order, the constant by which income is multiplied can be factorized in the terms implying income (Equation 26). Thus, its presence mainly modifies the order of magnitude of the coefficients of these terms.

Finally, the estimated model is

$$\begin{cases} \alpha_{i} = 1 \Leftrightarrow [\log(W_{i} - \pi_{i}) - \log(W_{i})] + \tilde{p}_{o}[\log(W_{i}) - \log(W_{i} - L_{oi})] \\ + [qS_{i} + \theta E(Z_{aid,i})S_{i}][\log(W_{i}) - \log(W_{i} - \beta L_{oi})] + o_{t}O_{ti} + o_{l}O_{li} \\ + h_{c}H_{ci} + h_{w}H_{wi} + h_{d}H_{di} + h_{t}H_{ti} + \delta E(Z_{peer,i}) + b_{cl}B_{cli} + b_{a}B_{ai} + \nu\epsilon_{i} + \eta_{i} \ge 0, (26) \\ \text{if } \alpha_{i} = 1, \log(\pi_{i}) = c_{\pi} + y\log(Y_{i}) + n\log(N_{i}) + \log(1 - \tau O_{ti}) \\ - \log(1 - \kappa - \rho R_{i}) + \sigma\epsilon_{i}, \\ \text{if } \alpha_{i} = 0, \pi_{i} = 0. \end{cases}$$

$$(27)$$

where errors  $\epsilon$  and  $\eta$  follow independent centered normal distributions with unit variance,  $c_{\pi} = \log(cp_o l)$  and  $\rho = cpr/(1+r)$  are estimated parameters and  $\boldsymbol{\kappa} = ckr/(1+r)$ ,  $\tilde{\boldsymbol{p}}_{\boldsymbol{o}}$  and  $\boldsymbol{\beta}$  are calibrated parameters.

Identifying variables. Identification requires the presence of variables that explain the probability of purchasing insurance but not the insurance premium. These identifying variables are the dummies for houses still in construction  $(H_c)$  and without drainage  $(H_d)$ .<sup>57</sup> Economically, it means that houses still in construction and without drainage have a lower probability of being insured (because they are likely uninsurable) but, once a house is covered, the price of its coverage does not depend on these characteristics.

The model is overidentified, as identification requires to exclude one variable only from the demand equation. The two identifying variables are here compatible: when only one of the two is excluded from the premium, the remaining one is not significant in the premium equation and both variables are significant in the demand equation.

 $<sup>^{57}</sup>$ Houses still in construction and houses without drainage correspond to 3% and 53% of dwellings, respectively (Table 3). These dummies (as dummies for houses without hot water or toilets inside the dwelling) do not significantly explain the losses, even when considering that losses can be differently estimated by households and by insurers.

# 5 Results

Estimation is based on maximum likelihood and is detailed in Appendix A.

### 5.1 Supply

Insurance pricing. Table 5 presents the results of the estimation of the insurance premium (Equation 27). As expected, the insurance premium increases with respect to the standard of living (y > 0) and the number of rooms of the dwelling (n > 0), which are proxies for the insured value (furniture and building values). Besides, as tenants insure only a fraction of the total value of the dwelling, the insurance premium is lower for tenants  $(\tau > 0)$ . The premium increases with respect to the exposure to natural disasters  $(\rho > 0)$ , confirming that the potential loss of the insure of the exposure of his policyholders, even if it is to a very limited extent (Section 2).

Coefficient	Estimate	Standard error	$\Pr >  t value $
$c_{\pi}$	2.4	0.16	< 0.0001
y	0.22	0.016	$<\!0.0001$
n	0.32	0.047	$<\!0.0001$
au	0.29	0.027	$<\!0.0001$
ρ	0.056	0.0068	$<\!0.0001$
σ	0.61	0.015	$<\!0.0001$
$\kappa$	0.088	0	

Table 5: Estimation results: supply equation

Notes:  $\kappa = ckr/(1+r)$  is calibrated at 0.088 (using c = 1.3, k = 0.635 and r = 0.12). 2006 French Household Budget survey and GASPAR database. 2,809 observations.

**Insurance affordability.** Some overseas households may not afford insurance, as the median standard of living in the French overseas departments is almost 40% lower than the one in continental France (Michel et al., 2010). To determine whether insurance is affordable for overseas households, the premiums offered to the uninsured are here estimated. These premiums can be estimated using these coefficients.<sup>58</sup>

The premiums offered to the uninsured are on average 9% below the premiums paid by the insured, mainly because the uninsured are on average poorer (Table 6).<sup>59</sup> As the premium increases less than proportionally with respect to the income (y < 1, Table 5), the budget weight (ratio of the premium over the annual income) decreases with respect to the income: the budget weight of the premium is so higher for the uninsured (the mean being 2.1%) than for the insured (1.4%), while remaining limited (Table 6). This limited budget weight of insurance premium for the uninsured suggests that insurance premiums should not prevent them from purchasing insurance. Answering properly that question requires estimation of insurance demand and in particular of the elasticity of insurance demand with respect to the premium.

	Mean	Lower quartile	Median	Upper quartile
Unin	sured hou	$\iota seholds$		
Premium $(2006 \in)$	231	187	236	274
Annual income $(2006 \in)$	15,735	7,756	13,032	20,236
Budget weight	2.1%	1.2%	1.7%	2.6%
Inst	ured hous	eholds		
Premium $(2006 \notin)$	254	118	180	300
Annual income $(2006 \in)$	30,217	$13,\!974$	$25,\!056$	40,222
Budget weight	1.4%	0.5%	0.8%	1.4%

Table 6: Home insurance: premium and budget weight

*Notes:* 2006 French Household Budget survey and GASPAR database. 2,809 observations.

<sup>&</sup>lt;sup>58</sup>These estimated coefficients (Table 5) correct the presence of a significant selection bias in Equation 26 (Table 7). In other words, this estimation takes into account the presence of unobserved heterogeneities that increase the probability of purchasing insurance and the insurance premium. These unobserved heterogeneities may be relative to risk aversion: households with higher risk aversion have a higher probability to purchase insurance; their higher risk aversion may be partially measured by insurers and so reflected in their premium. Regarding residuals, using their estimated variance implies that residuals for the uninsured are assumed to have the same variance than the ones for the insured.

<sup>&</sup>lt;sup>59</sup>The elasticity of demand with respect to the income is estimated in Section 5.2.

### 5.2 Demand

Table 7 presents the estimation results for the demand equation (Equation 26). These results are now precisely commented and derived to quantify and compare the magnitude of demand determinants.

Coefficient	Estimate	Standard error	$\Pr >  t value $
$O_t$	0.34	0.070	< 0.0001
$o_l$	0.83	0.094	$<\!0.0001$
$h_c$	-0.71	0.23	0.0020
$h_w$	-0.85	0.076	$<\!0.0001$
$h_d$	-0.50	0.061	$<\!0.0001$
$h_t$	-0.70	0.20	0.00050
q	-0.065	0.011	$<\!0.0001$
$\theta$	0.095	0.020	$<\!0.0001$
δ	0.67	0.13	$<\!0.0001$
$b_{cl}$	0.77	0.11	$<\!0.0001$
$b_a$	-0.53	0.099	$<\!0.0001$
ν	0.41	0.095	$<\!0.0001$
$ ilde{p}_{o}$	0.075	0	
eta	15	0	

Table 7: Estimation results: demand equation

Notes:  $\tilde{p}_o$  and  $\beta$  are calibrated at 0.075 and 15, respectively. 2006 French Household Budget survey and GASPAR database. 2,809 observations.

Small elasticity of insurance demand with respect to premium. The elasticity of insurance demand with respect to the premium can be calculated from these results. The elasticity of insurance demand with respect to the premium is  $-5 \cdot 10^{-4}$ , which is far smaller than results found by other studies for home and flood insurance (Table 8).<sup>60</sup> When the premium increases by 50%, the number of households who are willing to purchase insurance decreases by only 0.2%. This small price elasticity is due to the subsidized natural disasters coverage provided by home insurance. This result confirms that overseas households are not deterred from purchasing insurance by its price.

<sup>&</sup>lt;sup>60</sup>This is not due to the fact that the premium is negligible with respect to households' holdings. Indeed, even when the model is estimated using annual income as wealth, the price elasticity of insurance demand remains low  $(-5 \cdot 10^{-2})$ .

Income elasticity of insurance demand. The elasticity of insurance demand with respect to the income can also be calculated and its order of magnitude is consistent with other studies (Table 8).<sup>61</sup> The income elasticity of insurance demand equals 0.10. Its positive sign confirms that the insured are on average richer than the uninsured (Table 6). Income elasticity of insurance demand may be positive or negative. Indeed, two opposite effects are at stake. On the one hand, theory predicts that, if the absolute risk aversion decreases with respect to the income, the demand for insurance also decreases with respect to the income (Schlesinger, 2000). On the other hand, wealthier households may buy costlier houses, exposing themselves this way to higher potential losses and increasing their need of coverage (Cleeton and Zellner, 1993).<sup>62</sup> Here a third effect is probably also at stake. Low income households likely benefit from more assistance after natural disasters,<sup>63</sup> which decreases insurance demand from low-income households. The positive sign of income elasticity means that the last two effects exceed the first one.

Line of insurance and place	Definition of demand	Price elasticity	Income elasticity	Citation
Home insurance				
French overseas departments	(PP)	$-5\cdot10^{-4}$	0.10	Current study
Florida	(FA)	-1.08	0.06	Grace et al.
New York	(FA)	-0.86	-0.03	(2004)
National flood insurance				
Unites States	(PP)	-0.11	1.40	Browne and
Unites States	(FA)	-1.00	1.51	Hoyt $(2000)$

Table 8: Price and income elasticities of demand for home and flood insurance

*Notes:* insurance demand is defined either by the percentage of purchased policies in the population (PP) or by the face amount of coverage (FA).

**Insurance obligations.** Tenants and even more homeowners with outstanding loans have a higher probability of purchasing insurance than homeowners  $(o_l > o_t > o_t)$ 

 $<sup>^{61}</sup>$ Given that wealth is proportional to income (Section 4), elasticities with respect to wealth or to income are identical.

<sup>&</sup>lt;sup>62</sup>Cleeton and Zellner (1993) show that the income elasticity of insurance demand is positive if  $\phi_a + \eta > 1$ , where  $\phi_a$  is the elasticity of relative risk aversion to initial income and  $\eta$  is the elasticity of the amount of risk with respect to the initial income.

<sup>&</sup>lt;sup>63</sup>For example financial assistance by the rescue fund for overseas decreases with respect to the income.

0). This result shows that the existing constraints relative to insurance purchase are operant. Besides, they have an important impact: if all households were tenants, the percentage of insured households would go from 48% (Table 3) to 60% (Table 9); if all households were homeowners with outstanding loans, the percentage of insured households would reach 72% (Table 9).<sup>64</sup>

Uninsurable housing. As expected, households living in a house in construction or without modern conveniences have a smaller probability of purchasing insurance  $(h_c, h_w, h_d, h_t > 0)$ . In practice, insurers can control building quality and permit, either before selling the contract or once a loss has occurred before paying compensation. In any case, this control can be easily anticipated by households. The impact of uninsurable housing is important: if all households were living in a house still in construction, the percentage of insured households would go from 48% down to 19%; if all dwellings were houses without hot water, the insurance penetration rate would go down to 13%; if they were living in a house without drainage, this rate would go down to 36%; if all dwellings were houses without toilets inside the building, this rate would go down to 19% (Table 9).

Table 9: Impact of uninsurable housing and insurance obligations on insurance demand

Assumption	Percentage of insured households
$O_t = 1$	60%
$O_l = 1$	72%
$H_c = 1$	19%
$H_w = 1$	13%
$H_d = 1$	36%
$H_t = 1$	19%

*Notes:* the initial percentage of insured households is 48%. 2006 French Household Budget survey and GASPAR database. 2,809 observations.

<sup>&</sup>lt;sup>64</sup>Purchasing home insurance is also required as a condition for obtaining a mortgage in the United States (Browne and Hoyt (2000), Kunreuther and Pauly (2006)) (Section 2). Browne and Hoyt (2000) show that the number of mortgages per capita in the United States is negatively related to the number of policies purchased per capita, likely because the level of mortgages captures wealth and income effects.

**Charity hazard.** The probability of purchasing insurance decreases with respect to the number of past disasters that have occurred in the municipality. As explained in Section 4, the negative sign of q reveals the presence of charity hazard that outweighs the utility loss effect, and means that anticipated assistance by households increases with respect to the number of past disasters that have occurred in the municipality. There is indeed a cumulative effect in the anticipation of assistance: households living in municipalities where numerous disasters occurred have noticed the importance of assistance, probably more than the other households have; therefore they anticipate higher amounts of *ex post* aid.

A vicious circle of uninsurance. The penetration rate in the neighborhood increases the individual probability of purchasing insurance ( $\delta > 0$ ), which reveals peer effects: the more neighbors are insured, the higher is the individual probability of purchasing insurance. This peer effect is significant at the municipal level, but not when defining the group of peers as households sharing same observed characteristics inside a municipality.

Besides, the penetration rate in the group for aid eligibility decreases the charity hazard effect ( $\theta > 0$ ): the percentage of insured households around one individual decreases his likelihood to get assistance after a disaster. The relevant group for aid eligibility is also the municipality ( $J_{aid} = J_{peer}$ ). This suggests that there is no favoritism towards households sharing one of the observed characteristics.<sup>65</sup>

Assuming that 3 over 4 households living in the municipality were insured, if there were peer effects only, the individual probability of purchasing insurance would reach 0.65; if the endogenous nature of assistance only was at stake, this probability would reach 0.49 (Table 10).

The place of birth explains the probability of purchasing insurance when this characteristic is simply added to the demand equation. Indeed, all things being equal,

 $<sup>^{65}</sup>$ The two external effects of neighbors' insurance decision (based on peer effect or aid eligibility) remain significant when considering one without the other.

Assumption	Individual probability of purchasing insurance
Municipal insurance rate $= 75\%$	
via peer effects only	0.65
via aid eligibility only	0.49

Table 10: Impact of the municipal insurance rate

*Notes:* the initial probability of purchasing insurance is 0.48. 2006 French Household Budget survey and GASPAR database. 2,809 observations.

households whose reference person is born in continental France have a higher probability of purchasing insurance  $(b_{cl} > 0)$ , whereas households born abroad have a lower probability of purchasing insurance  $(b_a < 0)$ . This result suggests that having grown up in a place where a wide majority of people are insured increases the probability of being insured. This "initial peer effect" has also an important magnitude. If all households were born in continental France, the percentage of insured households would go from 48% to 71%. On the contrary, if all households were born abroad, the percentage of insured households would go from 48% down to 29% (Table 11).

Table 11: Impact of the place of birth on insurance demand

Assumption	Percentage of
	insured households
$B_{cl} = 1$	71%
$B_a = 1$	29%

*Notes:* the initial percentage of insured households is 48%. 2006 French Household Budget survey and GASPAR database. 2,809 observations.

Therefore households are not deterred from purchasing insurance by relatively high insurance premiums but by assistance provided after disasters; uninsurable housing also decreases the probability of being insured. My findings also suggest that the neighbors' insurance choices impact the individual insurance decision via peer effects and via neighborhood eligibility for assistance.

# 6 Discussion

The French overseas departments provide a rare natural experiment of a welldeveloped natural disasters insurance supply in Latin America, the Caribbean and other exposed small island countries. The determinants of insurance coverage on the demand side are uninsurable housing, which mainly applies in developing countries, and charity hazard, which also widely applies in developed countries. These two phenomena are here precisely discussed.

### 6.1 Uninsurable housing

Uninsurable housing widely applies in developing countries (Gilbert, 2001) and is well-documented in Latin America and the Caribbean (Section 2). Many developing countries (located in Africa, Asia and Pacific region, Europe or Middle East) benefit from some aids by the World Bank specifically dedicated to dwellings repair or rebuilding (Gilbert, 2001). These reconstruction projects often include an improvement of dwelling quality (introduction or use of earthquake resistant materials and designs, training of local masons, carpenters and artisans) (Gilbert, 2001). In the French overseas departments, building aid is already in place (Tjibaou, 2004).<sup>66</sup> This housing policy contributes to the decrease of uninsurable housing (Table 12). This may probably partly explain why the penetration rate has been progressively increasing (except in French Guiana, where uninsurable housing remains especially important) since 1995 (Table 13), since the impact of uninsurable housing on insurance demand is important (Section 5, Table 9).

### 6.2 Charity hazard

Charity hazard has an important magnitude in many developing countries (Gilbert, 2001) and among them the Caribbean region (Section 2); some aids, for example

 $<sup>^{66}</sup>$ Furthermore, recent legal evolutions enable homeowners of squalid dwellings with neither right nor title to be compensated if public operations require their dwelling to be demolished. See law n<sup>0</sup>2011-725 of June 23, 2011 relative to informal housing districts and fight against bad housing on overseas departments and regions.

Share of $(\%)$	Permanent	Dwellings	Traditional	Makeshift
in 1999 / in 2007	structures	in wood	huts	dwellings
French Guiana	$68.0 \ / \ 73.0$	16.8 / 16.4	$10.3 \ / \ 6.5$	4.8 / 4.2
Guadeloupe	$74.8 \ / \ 89.6$	$10.1\ /\ 5.5$	$12.6 \ / \ 3.6$	$2.5 \ / \ 1.2$
Martinique	$88.5 \ / \ 93.7$	$5.3 \; / \; 3.6$	$4.4 \ / \ 1.1$	$1.8\ /\ 1.7$
Réunion	$73.7 \ / \ 86.2$	$10.3 \ / \ 4.2$	$14.0 \ / \ 8.5$	$2.1 \ / \ 1.1$

Table 12: Evolution of dwelling quality in the French overseas departments

*Notes:* Main homes only are considered. Dwelling can be a house or an apartment. Population census by INSEE in 1999 and 2007.

Table 13: Evolution of home insurance penetration rate in the French overseas departments

(%)	1995	2001	2006
French Guiana	47	38	52
Guadeloupe	29	32	44
Martinique	39	41	50
Réunion	29	45	59

*Notes:* French Household Budget survey by INSEE in 1995, 2001 and 2006. 2 922 observations in 1995, 3 302 in 2001, 3 134 in 2006.

ones from the World Bank, are specifically dedicated to dwellings repair or rebuilding in developing countries (Gilbert, 2001). Charity hazard is also at stake in developed countries (see Raschky and Weck-Hannemann (2007) for a review).

Many European countries (Austria, Czech Republic, Germany, Italy, Poland, Slovakia) combine public assistance and private insurance with a low penetration rate (Maccaferri et al., 2012).<sup>67</sup> For example, in Canada, public assistance is also developed (through the Disaster Financial Assistance Arrangements and local funds created by some provinces) and Canadian households do not distinguish between public aids and compensations provided by insurers (Dumas et al., 2005). In Germany and in Italy, insurance is private and governmental assistance to flood victims is provided on an ad hoc basis; less than 10% of German households and about 5%

<sup>&</sup>lt;sup>67</sup>In all these countries that combine public assistance and private insurance, it is difficult to determine the causality between the development of public assistance and the low penetration rate of private insurance: was demand for private insurance reduced because of public aid? Or was assistance initially developed to make up for a limited private insurance supply?

of Italian buildings are insured against floods (Bouwer et al. (2007), Schwarze and Wagner (2007)). These few examples illustrate differences in the institutional design of governmental relief programs between countries. This design - actually more its transparency than the coverage magnitude - significantly determines the demand for private natural hazard insurance (Raschky et al., 2010).

Charity hazard may also occur in developed countries where public assistance coexists with public insurance. In the United States, flood insurance is offered by the Federal State and is purchased by a minority of households (Dixon et al. (2006), Kunreuther (1984)).<sup>68</sup> Before Hurricane Katrina, Browne and Hoyt (2000) and Kunreuther and Pauly (2006) show that a key explanation for the low demand for natural disasters insurance from American households is their biased risk perception and not charity hazard.<sup>69</sup> After Hurricane Katrina, Bush administration committed to provide billions of dollars in disaster relief to victims; this may have induced expectation of Federal assistance (Kunreuther and Pauly, 2005). Petrolia et al. (2012) show that the decision to purchase a flood policy is positively correlated with the eligibility for disaster assistance.

To what extent is charity hazard an issue? After all, as recalled by Raschky and Weck-Hannemann (2007), a catastrophe fund is *de facto* a "mandatory insurance". Indeed, one can argue that public assistance is not that much different from insurance subsidy: public assistance is a cross-subsidization from less exposed taxpayers to more exposed ones; similarly, insurance subsidy is a cross-subsidization from less exposed insured households to more exposed ones. This comparison is especially relevant for countries where insurance pricing implies insurance subsidy, such as France or the United States. Indeed, in France, the natural disasters pre-

 $<sup>^{68}</sup>$ In the United States, flood insurance, which is offered by the Federal State to households, is purchased by around half of the single-family homes living in special flood hazard areas - i.e. zones with a 100-year recurrence interval for flood - and by only 1% of single-family homes outside (Dixon et al., 2006).

<sup>&</sup>lt;sup>69</sup>For example, Browne and Hoyt (2000) test the presence of charity hazard and find a positive correlation between governmental aid and flood insurance purchase - and not a negative one. Their interpretation is that flood exposure may increase both governmental aid and insurance purchase.

mium is a fixed share of the home insurance premium (Section 2). In the United States, flood insurance is actuarial with subvention of specific risks and 22% of flood insurance policies are subsidized (Hayes and Neal, 2009).

Coate (1995) answers this very precise objection: compensation provided by insurance is defined ex ante, whereas compensation provided by aid is often defined expost. This main difference has two important consequences, both underlined by Coate (1995).

First, *ex post* assistance is likely to be inefficient. There are two main reasons to expect that people who provide assistance will not choose the optimal level of assistance. The first reason is that assistance may rely on approximate loss assessment or even on discretionary decisions. In the United States, half of disaster payments by the Federal Emergency Management Agency are politically motivated (Garrett and Sobel, 2003).<sup>70</sup> The second reason is that the uninsured can free-ride, since natural disasters assistance is provided via different channels. To that respect, the assistance providers themselves can consider that the level of assistance is not optimal.

Second, providing *ex post* assistance reduces self-responsibility and gives no incentive for prevention. It does not refrain households from living in exposed areas or from building vulnerable houses, while these choices increase future losses and so future assistance provided by the whole society. Certainly, insurance subsidy also reduces self-responsibility, but this subsidy can be made temporary or combined with other incentives for prevention. For example, in the United States, this subsidy is temporary: flood insurance is provided at subsidized rates until the completion of the community's flood rate map. In France, this subsidy goes with incentives for prevention: the natural disasters insurance deductible increases with respect to the

<sup>&</sup>lt;sup>70</sup>Similarly, in Pakistan, after the 2001 flood in Islamabad and Rawalpindi, public support checks were mainly distributed to family members and political supporters of local councilors who coordinate governmental assistance (Mustafa, 2003).

number of past disasters that have occurred in the municipality;<sup>71</sup> increasing the premium with respect to the risk exposure could also be considered.<sup>72</sup> Efficiency of such insurance policies clearly requires that the most exposed households had purchased insurance.

A third argument can be added to Coate (1995)'s ones: public assistance may distort the fiscal system and so redistribution between the rich and the poor. For example, in the United States, after Hurricane Andrew in 1992, assistance to the rich was funded by the poor, as Federal assistance was counterbalanced by a reduction of social budget (Favier and Pfister, 2007). On the contrary, in the French overseas departments, low income households benefit from more assistance after natural disasters (for example via the rescue fund for overseas). Actually this precisely contributes to explain why the uninsured are the poor in these departments (Subsection 5.2).

# 7 Conclusion

This paper studies the reasons for underinsurance against natural disasters in highly exposed areas. Limited insurance supply is commonly identified as a primary factor causing low insurance coverage in exposed areas. The French overseas departments provide a rare situation of a well-developed natural disasters insurance supply in highly exposed regions. Indeed, the French natural disasters insurance system is guaranteed by the French government; first foreseen for continental France only, it was extended to overseas departments after Hurricane Hugo in 1989, in a state of emergency. This natural experiment enables to analyze the determinants of insurance coverage on the demand side.

<sup>&</sup>lt;sup>71</sup>The natural disasters insurance deductible paid by individuals is fixed by the government and can increase in municipalities which suffered several natural disasters and made however no risk prevention plan (Insurance Code, section L. 125-1, annex I). As the very wide majority of the municipalities in the French overseas departments have already undertaken or set up such plans (Section 2), this rule has a negligible impact in these departments.

<sup>&</sup>lt;sup>72</sup>For now, increasing the premium with respect to the risk exposure is considered for insurance of firms' or local authorities' buildings only, not for home insurance. See http://www.senat.fr/leg/pjl11-491.html.

Using unique household-level micro-data about the insured and the uninsured, I estimate a semi-structural model of equilibrium on insurance market which had not been empirically tested. The structural approach enables to measure distortions on natural risk pricing due to insurance supply regulation and to disentangle the different possible causes of underinsurance on the demand side. I show that under-insurance in the French overseas departments is neither due to perception biases nor to unaffordable insurance, but mainly to uninsurable housing and to anticipated assistance, which crowds out insurance. Besides, the neighbors' insurance choices impact the individual insurance decision via peer effects and via neighborhood eligibility for assistance. Finally, I show that the existing insurance obligations (*de facto* for homeowners with outstanding loans, as in most Caribbean countries, and *de jure* for French tenants) are operant but do not guarantee targeted households to be insured, as they may not renew their insurance contracts once they have settled in.

There are two substantive lessons that one learns from this analysis. First, the main reasons for the low demand for insurance coverage against natural disasters in exposed areas are uninsurable housing, which mainly applies in developing countries, and charity hazard, which also widely applies in developed countries. Second, and consequently, these findings suggest that the development of an affordable supply of natural disasters coverage would probably increase the insurance penetration rate in disaster-prone areas, but would unlikely imply a wide majority of insured households not only because the social equilibrium of uninsurance has to be broken, but also because of charity hazard and, in developing countries, because of uninsurable housing. Thus, the development of a natural disasters coverage supply in disaster-prone areas (either via governmental initiatives or via microinsurance) would unlikely ensure the ability of governments to massively transfer catastrophic risk via coverage mechanisms if it does not go with a policy reducing charity hazard and, in developing countries, uninsurable housing.

# References

- Abel, A., 1986. Capital Accumulation with Adverse Selection and Uncertain Lifetimes. Econometrica 54, 1079–97.
- Analytica, O., 1997. Caribbean: Catastrophe Risk. September 3, 1997. Technical Report. Oxford Analytica Brief.
- Arvan, L., Nickerson, D., 2006. Private Investment, Public Aid and Endogenous Divergence in the Evolution of Urban Neighborhoods. Journal of Real Estate Finance and Economics 32, 83–100.
- Auffret, P., 2003. Catastrophe Insurance Market in the Caribbean Region: Market Failures and Recommendations for Public Sector Interventions. Working Paper 2963. World Bank Policy Research.
- Barnett, B.J., Barrett, C.B., Skees, J.R., 2008. Poverty Traps and Index-Based Risk Transfer Products. World Development 36, 1766–1785.
- Barredo, J.I., 2009. Normalised Flood Losses in Europe: 1970-2006. Natural Hazards and Earth System Sciences 9, 97–104.
- Bevere, L., Rogers, B., Grollimund, B., 2011. Sigma Report: Natural Catastrophes and Man-made Disasters in 2010: A Year of Devastating and Costly Events. Technical Report. Swiss Re.
- Bidan, P., 2000. Indemnisation des Catastrophes Naturelles : de la Naissance vers l'Age Adulte. Risques Numéro 42.
- Borensztein, E., Cavallo, E., Valenzuela, P., 2009. Debt Sustainability under Catastrophic Risk: The Case for Government Budget Insurance. Risk Management and Insurance Review 12, 273–294.
- Bouwer, L.M., Huitema, D., Aerts, J.C., 2007. Adaptive Flood Management: The Role of Insurance and Compensation in Europe. Technical Report. 2007 Amsterdam Conference - Conference on the Human Dimensions of Global Environmental Change, 24-26 May 2007, Amsterdam, The Netherlands.
- Brown, D.P., 1990. Age Clienteles Induced by Liquidity Constraints. International Economic Review 31, 891–912.
- Brown, J.R., Finkelstein, A., 2008. The Interaction of Public and Private Insurance: Medicaid and the Long-Term Care Insurance Market. American Economic Review 98, 1083–1102.
- Browne, M.J., Hoyt, R.E., 2000. The Demand for Flood Insurance: Empirical Evidence. Journal of Risk and Uncertainty 20:3, 291–306.
- Calvet, L., Grislain-Letrémy, C., 2011. L'Assurance Habitation dans les Départements d'Outre-mer : Une Faible Souscription. Economie et Statistique Numéro 447, 57–70.
- Castéran, B., Ricroch, L., 2008. Les Logements en 2006. Le Confort S'Améliore, Mais Pas Pour Tous. INSEE Première Numéro 1202.

- Cavallo, E.A., Noy, I., 2009. The Economics of Natural Disasters: A Survey. Inter-American Development Bank Working Paper 35.
- Charvériat, C., 2000. Natural Disasters in Latin America and the Caribbean: An Overview of Risk. Inter-American Development Bank Working Paper 434.
- Chernew, M., Cutler, D., Keenan, P.S., 2005. Charity Care, Risk Pooling, and the Decline in Private Health Insurance. American Economic Review Papers and Proceedings of the One Hundred Seventeenth Annual Meeting of the American Economic Association, January 7-9, 2005 95, 209–213.
- Chiappori, P.A., Salanié, B., 2008. Modeling Competition and Market Equilibrium in Insurance: Empirical Issues. American Economic Review: Papers & Proceedings 98, 146–150.
- Cleeton, D.L., Zellner, B.B., 1993. Income, Risk Aversion, and the Demand for Insurance. Southern Economic Journal 60, 146–156.
- Coate, S., 1995. Altruism, the Samaritan's Dilemma, and Government Transfer Policy. American Economic Review 85, 46–57.
- Deryugina, T., Kirwan, B., 2012. Charity Hazard in Crop Insurance. Preliminary draft .
- DIREN, 2005. Tableau de Bord de l'Environnement, Martinique 2005. Technical Report. Direction Régionale de l'Environnement Martinique.
- Dixon, L., Clancy, N., Seabury, S.A., Overton, A., 2006. The National Flood Insurance Program's Market Penetration Rate. Technical Report. RAND Corporation.
- Dumas, P., Chavarot, A., Legrand, H., Macaire, A., Dimitrov, C., Martin, X., Queffelec, C., 2005. Rapport Particulier sur la Prévention des Risques Naturels et la Responsabilisation des Acteurs. Mission d'Enquête sur le Régime d'Indemnisation des Victimes de Catastrophes Naturelles. French Treasury General Inspection, French Environment General Inspection and French General Council of Civil Engineering.
- Erhard-Cassegrain, A., Massé, E., Momal, P., 2006. Evolution du régime d'indemnisation des catastrophes naturelles. Document de travail de la Direction des Etudes Economiques et de l'Evaluation Environnementale, Série Synthèses Numéro 04-S06.
- Evans, A.F., 1996. Insurance, Reinsurance and Catastrophe Protection in the Caribbean. Technical Report. Organization of American States General Secretariat. Working Paper prepared in collaboration with the World Bank.
- Favier, R., Larhra, U., 2007. From Solidarity to Individual Compensation: Assistance Mechanisms Faced with the Emergence of Liberalism in France during the 18th Century. In Solidarité et Assurance : Les Sociétés Européennes Face aux Catastrophes (17e-21e s.). CNRS MSH-Alpes. pp. 57–79.
- Favier, R., Pfister, C., 2007. Introduction. In Solidarité et Assurance : Les Sociétés Européennes Face aux Catastrophes (17e-21e s.). CNRS MSH-Alpes. pp. 9–13.

- Fay, M., Wellenstein, A., 2005. The Urban Poor in Latin America. World Bank. chapter Keeping a Roof over One's Head: Improving Access to Safe and Decent Shelter. pp. 91–124.
- FFSA, 2006. Présentation de la Fédération Française des Sociétés d'Assurance au Séminaire MultiRisques Habitation du 16 mai 2006. http://www.ffsa.fr/sites/upload/reprise/docs/application/pdf/2010-02/apmrhmai06.pdf. Technical Report.
- Forgeot, G., Celma, C., 2009. Les Inégalités aux Antilles Guyane : Dix Ans d'Evolution. Technical Report. Institut National de la Statistique et des Etudes Economiques (INSEE) and Caisses d'Allocations Familiales d'Antilles-Guyane.
- Freeman, P.K., Keen, M., Manil, M., 2003. Dealing with Increased Risk of Natural Disasters: Challenges and Options. Technical Report. International Monetary Fund Working Paper No. 03/197.
- Garnesson, C., Hecquet, V., 2007. Le Logement Dans les DOM. Technical Report 67. Antiane.
- Garrett, T.A., Sobel, R.S., 2003. The Political Economy of FEMA Disaster Payments. Economic Inquiry 41, 496–509.
- Gaschen, S., Hausmann, P., Menzinger, I., Schaad, W., 1998. Floods An Insurable Risk? A Market Survey. Technical Report. Swiss Re.
- Gilbert, R., 2001. Doing More for Those Made Homeless by Natural Disasters. Technical Report. World Bank.
- Gollier, C., 2003. To Insure or Not to Insure?: An Insurance Puzzle. Geneva Papers on Risk and Insurance - Theory 28, 5–24.
- Gollier, C., 2007. Comment Intégrer le Risque Dans le Calcul Economique ? Revue d'économie politique 117, 209–223.
- Grace, M.F., Klein, R.W., Kleindorfer, P.R., 2004. Homeowner's Insurance with Bundled Catastrophe Coverage. Journal of Risk and Insurance 71, 351–379.
- Grislain-Letrémy, C., Peinturier, C., 2010. Le Régime des Catastrophes Naturelles en France Métropolitaine. Etudes et Documents, Ministère de l'Ecologie, de l'Energie, du Développement Durable et de l'Aménagement du Territoire 22.
- Halek, M., Eisenhauer, J.G., 2001. Demography of Risk Aversion. Journal of Risk and Insurance 68, 1–24.
- Hayes, T.L., Neal, D.A., 2009. Actuarial Rate Review, National Flood Insurance Program. Technical Report. Federal Emergency Management Agency.
- Heger, M., Julca, A., Paddison, O., 2008. Analysing the Impact of Natural Hazards in Small Economies: The Caribbean Case. Technical Report. United Nations University - World Institute for Development Economics Research. Research Paper No. 2008/25.
- Hernández-Murillo, R., Sengupta, R., 2012. The Effect of Neighborhood Spillovers on Mortgage Selection. Preliminary draft .

- Herring, B., 2005. Does Access to Charity Care for the Uninsured Crowd Out Private Health Insurance Coverage? Journal of Health Economics 24, 225–252.
- IDB (Ed.), 2000. Social Protection for Equity and Growth. Washington, DC, United States: Inter-American Development Bank. chapter Protecting the Poor Against Natural Disasters. pp. 47–76.
- IPCC, 2007. Fourth Assessment Report: Climate Change. Technical Report. Intergovernmental Panel on Climate Change.
- Jianakoplos, N.A., Bernasek, A., 1998. Are Women More Risk Averse? Economic Inquiry 36, 620–630.
- Joskow, P., Tirole, J., 2006. Retail Electricity Competition. RAND Journal of Economics 37, 799–815.
- Kreimer, A., Arnold, M., Barham, C., Freeman, P., Gilbert, R., Krimgold, F., Lester, R., Pollner, J.D., Vogt, T., 1999. Managing Disaster Risk in Mexico: Market Incentives for Mitigation Investment, The World Bank, Washington, D.C. Technical Report. World Bank.
- Kunreuther, H., 1984. Causes of Underinsurance Against Natural Disasters. Geneva Papers on Risk and Insurance 9, 206–220.
- Kunreuther, H., Novemsky, N., Kahneman, D., 2001. Making Low Probabilities Useful. Journal of Risk and Uncertainty 23, 103–120.
- Kunreuther, H., Pauly, M., 2004. Neglecting Disaster: Why Don't People Insure Against Large Losses? Journal of Risk and Uncertainty 28, 5–21.
- Kunreuther, H., Pauly, M., 2005. Insurance Decision-Making and Market Behavior. Foundations and Trends in Microeconomics 1, 63–127.
- Kunreuther, H., Pauly, M., 2006. Rules Rather than Discretion: Lessons from Hurricane Katrina. Journal of Risk and Uncertainty 33, 101–116.
- Landry, C.E., Jahan-Parvar, M.R., 2011. Flood Insurance Coverage in the Coastal Zone. Journal of Risk and Insurance 78, 361–388.
- Laroque, G., Salanié, B., 2002. Labour Market Institutions and Employment in France. Journal of Applied Econometrics 17, 25–48.
- Letrémy, C., 2009. Le Rôle de l'Assurance Dans la Prévention des Catastrophes Naturelles. Le Point Sur, Economie et Evaluation, Ministère de l'Ecologie, de l'Energie, du Développement Durable et de l'Amènagement du Territoire 9.
- Levin, I.P., Snyder, M.A., Chapman, D.P., 1988. The Interaction of Experiential and Situational Factors and Gender in a Simulated Risky Decision-Making Task. Journal of Psychology 122, 173–181.
- Lin, Y.S., 2005. Estimations of the Probability of Fire Occurrences in Buildings. Fire Safety Journal 40, 728–735.
- Maccaferri, S., Cariboni, F., Campolongo, F., 2012. Natural Catastrophes: Risk Relevance and Insurance Coverage in the EU. Technical Report. European Commission, Joint Research Centre, Scientific Support to Financial Analysis Unit, Institute for the Protection and Security of the Citizens.

- Mechler, R., Linnerooth-Bayer, J., Peppiatt, D., 2006. Microinsurance for Natural Disaster Risks in Developing Countries. Benefits, Limitations and Viability. A ProVention/IIASA study. Draft for discussion.
- Michel, C., Theulière, M., Missègue, N., 2010. Les Inégalités de Revenus entre les DOM et la Métropole. INSEE Première Numéro 1279.
- Morin, R.A., Suarez, A.F., 1983. Risk Aversion Revisited. The Journal of Finance 38, 1201–1216.
- MunichRe, 2012. TOPICS GEO Natural Catastrophes 2012. Analyses, Assessments, Positions. Technical Report.
- Mustafa, D., 2003. Reinforcing Vulnerability? Disaster Relief, Recovery and Response to the 2001 Flood in Rawalpindi, Pakistan. Environmental Hazards 5, 71–82.
- Niel, X., Beaumel, C., 2010. Le Nombre de Décès Augmente, l'Espérance de Vie Aussi. INSEE Première 1318.
- Office, U.G.A., 1983. The Effect of Premium Increases on Achieving the National Flood Insurance Program's Objectives. Technical Report RCED-83-107.
- Olive, V., Riviere, F., 2010. L'Habitat Dans les Outre-mer Français : Progrès, Enjeux, Disparités. Technical Report. Institut d'Emission des Départements d'Outre Mer, Institut d'Emission d'Outre Mer.
- Outreville, J.F., 2000. Handbook of Insurance. chapter The Retention Capacity of Insurance Markets in Developing Countries. pp. 749–766.
- Palsson, A.M., 1996. Does the Degree of Relative Risk Aversion Vary with Household Characteristics? Journal of Economic Psychology 17, 771–787.
- Pauly, M.V., 1974. Overinsurance and Public Provision of Insurance: The Roles of Moral Hazard and Adverse Selection. Quarterly Journal of Economics 88, 44–62.
- Pelling, M., Uitto, J.I., 2001. Small Island Developing States: Natural Disaster Vulnerability and Global Change. Environmental Hazards 3, 49–62.
- von Peter, G., von Dahlen, S., Saxena, S., 2012. Unmitigated Disasters? New Evidence on the Macroeconomic Cost of Natural Catastrophes. BIS Working Papers 394.
- Petrolia, D.R., Landry, C.E., Coble, K.H., 2012. Risk Preferences, Risk Perceptions, and Flood Insurance.
- Pollner, J., 2000. The Management of Catastrophic Risks using Pooled Insurance Structures and Alternative Financing & Risk Transfer Mechanisms. Technical Report. World Bank.
- Powell, M., Ansic, D., 1997. Gender Differences in Risk Behaviour in Financial Decision-Making: An Experimental Analysis. Journal of Economic Psychology 18, 605–628.

- Raschky, P.A., Schwarze, R., Schwindt, M., Zahn, F., 2010. Uncertainty of Governmental Relief and the Crowding out of Insurance. Monash University Discussion paper 05/10.
- Raschky, P.A., Weck-Hannemann, H., 2007. Charity Hazard A Real Hazard to Natural Disaster Insurance? Environmental Hazards 7, 321–329.
- Rasmussen, T.N., 2004. Macroeconomic Implications of Natural Disasters in the Caribbean. International Monetary Fund Working Paper WP/04/224.
- Rothschild, M., Stiglitz, J., 1976. Equilibrium in Competitive Insurance Markets: An Essay on the Economics of Imperfect Information. Quarterly Journal of Economics 90, 629–649.
- Schlesinger, H., 2000. The Theory of Insurance Demand. G. Dionne, Handbook of Insurance. chapter 5. pp. 131–151.
- Schubert, R., Brown, M., Gysler, M., Brachinger, H.W., 1999. Financial Decision-Making: Are Women Really More Risk-Averse? American Economic Review 89, 381–385.
- Schwarze, R., Wagner, G.G., 2007. The Political Economy of Natural Disaster Insurance: Lessons from the Failure of a Proposed Compulsory Insurance Scheme in Germany. European Environment 17, 403–415.
- Senate, F., 2005.Session  $\operatorname{at}$ the French Senate on Febru-17,2005.Comprehensive report of debates. Senate ary French http://www.senat.fr/seances/s200502/s20050217/s20050217002.html.
- Short, M.B., D'Orsogna, M.R., Brantingham, P.J., Tita, G.E., 2009. Measuring and Modeling Repeat and Near-Repeat Burglary Effects. Journal of Quantitative Criminology 25, 325–339.
- Tallon, J.M., Vergnaud, J.C., 2007. Incertitude en Economie de l'Environnement. Université Paris 1 Panthéon-Sorbonne (Post-Print and Working Papers). HAL.
- Tjibaou, M.C., 2004. Le Logement Dans l'Outre-mer Français. Technical Report. Conseil Economique et Social.
- Tversky, A., Kahneman, D., 1973. Availability: A Heuristic for Judging Frequency and Probability. Cognitive Psychology 5, 207–232.
- Tversky, A., Kahneman, D., 1981. The Framing of Decisions and the Psychology of Choice. Science, New Series 211, 453–458.
- WB, 1999. Proposed Strategy for a Small State Regional Catastrophe Insurance Program. Technical Report. World Bank, Caribbean Country Department.
- WB, 2011. Insuring Against Natural Disaster Risk in Mexico. Technical Report. World Bank, Treasury.
- Weitzman, M.L., 2009. On Modeling and Interpreting the Economics of Catastrophic Climate Change. The Review of Economics and Statistics 91, 1–19.

# A Appendix

Estimation is based on maximum loglikelihood. The calculation of the likelihood is hereafter detailed.

Recall that the estimated model is

$$\begin{cases} \alpha_{i} = 1 \Leftrightarrow \left[ \log(W_{i} - \pi_{i}) - \log(W_{i}) \right] + \tilde{p}_{o} \left[ \log(W_{i}) - \log(W_{i} - L_{oi}) \right] \\ + \left[ qS_{i} + \theta E(Z_{aid,i})S_{i} \right] \left[ \log(W_{i}) - \log(W_{i} - \beta L_{oi}) \right] + o_{t}O_{ti} + o_{l}O_{li} \\ + h_{c}H_{ci} + h_{w}H_{wi} + h_{d}H_{di} + h_{t}H_{ti} + \delta E(Z_{peer,i}) + b_{cl}B_{cli} + b_{a}B_{ai} + \nu\epsilon_{i} + \eta_{i} \ge 0, (28) \\ \text{if } \alpha_{i} = 1, \log(\pi_{i}) = c_{\pi} + y\log(Y_{i}) + n\log(N_{i}) + \log(1 - \tau O_{ti}) \\ - \log(1 - \kappa - \rho R_{i}) + \sigma\epsilon_{i}, \end{cases}$$

$$(29)$$

$$\text{if } \alpha_{i} = 0, \pi_{i} = 0.$$

where  $\boldsymbol{\kappa}$  is a calibrated parameter. I denote

$$Z_{\alpha} = \tilde{\boldsymbol{p}}_{\boldsymbol{o}}[\log(W_i) - \log(W_i - L_{oi})] + [qS_i + \theta E(Z_{\text{aid},i})S_i][\log(W_i) - \log(W_i - \boldsymbol{\beta}L_{oi})]$$

$$+ o_t O_t + o_l O_l + h_c H_c + h_w H_w + h_d H_d + h_t H_t + \delta E(Z_{\text{peer}}) + b_{cl} B_{cl} + b_a B_a,$$
(30)

$$Z_{\pi} = c_{\pi} + y \log(Y) + n \log(N) + \log(1 - \tau O_t) - \log(1 - \kappa - \rho R).$$
(31)

Besides, the probability density function of centered normal distribution with unit variance is denoted  $\varphi(\cdot)$  and the cumulative density function is denoted by  $\Phi(\cdot)$ .

For an insured household who pays a premium  $\pi$ , the probability of purchasing insurance can be directly calculated using (19). Using the symmetry of the normal distribution, I get

$$Pr\left(\eta \ge -\left(\log\left(1 - \frac{\pi}{W}\right) + Z_{\alpha} + \nu\epsilon\right)\right) = \Phi\left(\log\left(1 - \frac{\pi}{W}\right) + Z_{\alpha} + \nu\epsilon\right), \quad (32)$$

and the hazard is  $\epsilon = (\log(\pi) - Z_{\pi})/\sigma$  with probability  $1/\sigma \cdot \varphi ((\log(\pi) - Z_{\pi})/\sigma)$ .

Thus, for an insured household who pays a premium  $\pi$ , the likelihood function is

$$\frac{1}{\sigma}\varphi\left(\frac{\log(\pi) - Z_{\pi}}{\sigma}\right)\Phi\left(\log\left(1 - \frac{\pi}{W}\right) + Z_{\alpha} + \nu\frac{\log(\pi) - Z_{\pi}}{\sigma}\right),\tag{33}$$

For an uninsured household, the premium is not observed. Thus, the expected value of the probability of not purchasing insurance is<sup>73</sup>

$$1 - \int_{\mathbb{R}} \underbrace{\Phi\left(\log\left(1 - \frac{\exp(Z_{\pi} + \sigma\epsilon)}{W}\right) + Z_{\alpha} + \nu\epsilon\right)}_{F(\epsilon)} \varphi(\epsilon) d\epsilon.$$
(34)

I use the method exposed by Laroque and Salanié (2002) to approximate the integral that appears in the likelihood.<sup>74</sup> Following their estimation method, I denote by  $\epsilon_i$  the ith *m*-quantile ( $\Phi(\epsilon_i) = i/m$ ) and calculate  $\bar{\epsilon}_i$ , the average normal-weighted point in each interval [ $\epsilon_i, \epsilon_{i+1}$ ]. As  $x\phi(x) = -\phi'(x)$ ,

$$\bar{\epsilon}_i = \frac{\int_{\epsilon_i}^{\epsilon_{i+1}} x \phi(x) dx}{\Phi(\epsilon_{i+1}) - \Phi(\epsilon_i)} = m \bigg[ \phi(\epsilon_i) - \phi(\epsilon_{i+1}) \bigg],$$
(35)

and the integral can be approximated by

$$\int_{\mathbb{R}} F(\epsilon) \varphi(\epsilon) d\epsilon \approx \frac{1}{m} \sum_{i=0}^{m-1} F(\bar{\epsilon}_i).$$
(36)

Results are here presented for m = 10; they are robust when using m = 20.

<sup>&</sup>lt;sup>73</sup>The likelihood function for the uninsured takes into account the fact that the selection bias  $\nu \epsilon$  and the estimated premium for the uninsured both depend on the error term  $\epsilon$  (Equation 34).

<sup>&</sup>lt;sup>74</sup>Laroque and Salanié (2002) explain the wage and the participation decision on labor market, taking into account the fact that the decision to work depends on the wage. Their estimation is based on maximum likelihood and requires the approximation of a similar integral.

#### Liste des documents de travail de la Direction des Études et Synthèses Économiques

G 9001	J. FAYOLLE et M. FLEURBAEY Accumulation, profitabilité et endettement des entreprises
G 9002	H. ROUSSE Détection et effets de la multicolinéarité dans les modèles linéaires ordinaires - Un prolongement de la réflexion de BELSLEY, KUH et WELSCH
G 9003	P. RALLE et J. TOUJAS-BERNATE Indexation des salaires : la rupture de 1983
G 9004	D. GUELLEC et P. RALLE Compétitivité, croissance et innovation de produit
G 9005	P. RALLE et J. TOUJAS-BERNATE Les conséquences de la désindexation. Analyse dans une maquette prix-salaires
G 9101	Équipe AMADEUS Le modèle AMADEUS - Première partie - Présentation générale
G 9102	J.L. BRILLET Le modèle AMADEUS - Deuxième partie - Propriétés variantielles
G 9103	D. GUELLEC et P. RALLE Endogenous growth and product innovation
G 9104	H. ROUSSE Le modèle AMADEUS - Troisième partie - Le commerce extérieur et l'environnement international
G 9105	H. ROUSSE Effets de demande et d'offre dans les résultats du commerce extérieur manufacturé de la France au cours des deux dernières décennies
G 9106	B. CREPON Innovation, taille et concentration : causalités et dynamiques
G 9107	B. AMABLE et D. GUELLEC Un panorama des théories de la croissance endogène
G 9108	M. GLAUDE et M. MOUTARDIER Une évaluation du coût direct de l'enfant de 1979 à 1989
G 9109	P. RALLE et alii France - Allemagne : performances économiques comparées
G 9110	J.L. BRILLET Micro-DMS NON PARU
G 9111	A. MAGNIER Effets accélérateur et multiplicateur en France depuis 1970 : quelques résultats empiriques
G 9112	B. CREPON et G. DUREAU Investissement en recherche-développement : analyse de causalités dans un modèle d'accélé- rateur généralisé
G 9113	J.L. BRILLET, H. ERKEL-ROUSSE, J. TOUJAS- BERNATE "France-Allemagne Couplées" - Deux économies vues par une maquette macro-économétrique
G 9201	W.J. ADAMS, B. CREPON, D. ENCAOUA Choix technologiques et stratégies de dissuasion d'entrée
G 9202	J. OLIVEIRA-MARTINS, J. TOUJAS-BERNATE

Macro-economic import functions with imperfect competition - An application to the E.C. Trade G 9203 I. STAPIC Les échanges internationaux de services de la France dans le cadre des négociations multilatérales du GATT Juin 1992 (1ère version) Novembre 1992 (version finale) G 9204 P SEVESTRE L'économétrie sur données individuellestemporelles. Une note introductive H FRKEL-ROUSSE G 9205 Le commerce extérieur et l'environnement international dans le modèle AMADEUS (réestimation 1992) G 9206 N. GREENAN et D. GUELLEC Coordination within the firm and endogenous arowth G 9207 A. MAGNIER et J. TOUJAS-BERNATE Technology and trade: empirical evidences for the major five industrialized countries G 9208 B. CREPON, E. DUGUET, D. ENCAOUA et P. MOHNEN Cooperative, non cooperative R & D and optimal patent life G 9209 B. CREPON et E. DUGUET Research and development, competition and innovation: an application of pseudo maximum likelihood methods to Poisson models with heterogeneity G 9301 J. TOUJAS-BERNATE Commerce international et concurrence imparfaite : développements récents et implications pour la politique commerciale G 9302 Ch. CASES Durées de chômage et comportements d'offre de travail : une revue de la littérature G 9303 H. ERKEL-ROUSSE Union économique et monétaire : le débat économique G 9304 N. GREENAN - D. GUELLEC / G. BROUSSAUDIER - L. MIOTTI Innovation organisationnelle, dynamisme technologique et performances des entreprises G 9305 P. JAILLARD Le traité de Maastricht : présentation juridique et historique J.L. BRILLET G 9306 Micro-DMS : présentation et propriétés G 9307 JI BRILLET Micro-DMS - variantes : les tableaux G 9308 S. JACOBZONE Les grands réseaux publics français dans une perspective européenne G 9309 L. BLOCH - B. CŒURE Profitabilité de l'investissement productif et transmission des chocs financiers G 9310 J. BOURDIEU - B. COLIN-SEDILLOT

310 J. BOURDIEU - B. COLIN-SEDILLOT Les théories sur la structure optimale du capital : quelques points de repère

francaises : une évaluation empirique des théories de la structure optimale du capital L. BLOCH - B. CŒURÉ G 9312 Q de Tobin marginal et transmission des chocs financiers Équipes Amadeus (INSEE), Banque de France. G 9313 Métric (DP) Présentation des propriétés des principaux modèles macroéconomigues du Service Public G 9314 B CREPON - E DUGUET Research & Development, competition and innovation G 9315 B. DORMONT Quelle est l'influence du coût du travail sur l'emploi? G 9316 D. BLANCHET - C. BROUSSE Deux études sur l'âge de la retraite G 9317 D. BLANCHET Répartition du travail dans une population hétérogène : deux notes G 9318 D. EYSSARTIER - N. PONTY AMADEUS - an annual macro-economic model for the medium and long term G. CETTE - Ph. CUNÉO - D. EYSSARTIER -G 9319 J. GAUTIÉ Les effets sur l'emploi d'un abaissement du coût du travail des jeunes D. BLANCHET G 9401 Les structures par âge importent-elles ? G 9402 J GAUTIÉ Le chômage des ieunes en France : problème de formation ou phénomène de file d'attente ? Quelques éléments du débat G 9403 P. QUIRION Les déchets en France : éléments statistiques et économiques G 9404 D. LADIRAY - M. GRUN-REHOMME Lissage par moyennes mobiles - Le problème des extrémités de série V MAILLARD G 9405 Théorie et pratique de la correction des effets de jours ouvrables F. ROSENWALD G 9406 La décision d'investir G 9407 S. JACOBZONE Les apports de l'économie industrielle pour définir la stratégie économique de l'hôpital public G 9408 L. BLOCH, J. BOURDIEU. B. COLIN-SEDILLOT, G. LONGUEVILLE Du défaut de paiement au dépôt de bilan : les banquiers face aux PME en difficulté G 9409 D. EYSSARTIER, P. MAIRE Impacts macro-économigues de mesures d'aide au logement - quelques éléments d'évaluation F. ROSENWALD G 9410 Suivi conjoncturel de l'investissement C. DEFEUILLEY - Ph. QUIRION G 9411 Les déchets d'emballages ménagers : une

J. BOURDIEU - B. COLIN-SEDILLOT

Les décisions de financement des entreprises

G 9311

analyse économique des politiques française et allemande

- G 9412 J. BOURDIEU B. CŒURÉ -B. COLIN-SEDILLOT Investissement, incertitude et irréversibilité Quelques développements récents de la théorie de l'investissement
- G 9413 B. DORMONT M. PAUCHET L'évaluation de l'élasticité emploi-salaire dépendelle des structures de qualification ?
- G 9414 I. KABLA Le Choix de breveter une invention
- G 9501 J. BOURDIEU B. CŒURÉ B. SEDILLOT Irreversible Investment and Uncertainty: When is there a Value of Waiting?
- G 9502 L. BLOCH B. CŒURÉ Imperfections du marché du crédit, investissement des entreprises et cycle économique
- G 9503 D. GOUX E. MAURIN Les transformations de la demande de travail par qualification en France Une étude sur la période 1970-1993
- G 9504 N. GREENAN Technologie, changement organisationnel, qualifications et emploi : une étude empirique sur l'industrie manufacturière
- G 9505 D. GOUX E. MAURIN Persistance des hiérarchies sectorielles de salaires: un réexamen sur données françaises
- G 9505 D. GOUX E. MAURIN Bis Persistence of inter-industry wages differentials: a reexamination on matched worker-firm panel data
- G 9506 S. JACOBZONE Les liens entre RMI et chômage, une mise en perspective NON PARU - article sorti dans Économie et Prévision n° 122 (1996) - pages 95 à 113
- G 9507 G. CETTE S. MAHFOUZ Le partage primaire du revenu Constat descriptif sur longue période
- G 9601 Banque de France CEPREMAP Direction de la Prévision - Érasme - INSEE - OFCE Structures et propriétés de cinq modèles macroéconomiques français
- G 9602 Rapport d'activité de la DESE de l'année 1995
- G 9603 J. BOURDIEU A. DRAZNIEKS L'octroi de crédit aux PME : une analyse à partir d'informations bancaires
- G 9604 A. TOPIOL-BENSAÏD Les implantations japonaises en France
- G 9605 P. GENIER S. JACOBZONE Comportements de prévention, consommation d'alcool et tabagie : peut-on parler d'une gestion globale du capital santé ? Une modélisation microéconométrique empirique
- G 9606 C. DOZ F. LENGLART Factor analysis and unobserved component models: an application to the study of French business surveys
- G 9607 N. GREENAN D. GUELLEC La théorie coopérative de la firme

#### iii

- N. GREENAN D. GUELLEC G 9608 Technological innovation and employment reallocation
- Ph. COUR F. RUPPRECHT G 9609 L'intégration asymétrique au sein du continent américain : un essai de modélisation
- S. DUCHENE G. FORGEOT A. JACQUOT G 9610 Analyse des évolutions récentes de la productivité apparente du travail
- G 9611 X. BONNET - S. MAHFOUZ The influence of different specifications of wagesprices spirals on the measure of the NAIRU: the case of France
- G 9612 PH. COUR - E. DUBOIS, S. MAHFOUZ, J PISANI-FERRY The cost of fiscal retrenchment revisited: how strong is the evidence?
- G 9613 A. JACQUOT Les flexions des taux d'activité sont-elles seulement conioncturelles ?
- G 9614 ZHANG Yingxiang - SONG Xueging Lexique macroéconomique Français-Chinois
- J.L. SCHNEIDER G 9701 La taxe professionnelle : éléments de cadrage économique
- G 9702 J.L. SCHNEIDER Transition et stabilité politique d'un système redistributif
- G 9703 D. GOUX - E. MAURIN Train or Pay: Does it Reduce Inequalities to Encourage Firms to Train their Workers?
- P GENIER G 9704 Deux contributions sur dépendance et équité
- G 9705 E. DUGUET - N. IUNG R & D Investment, Patent Life and Patent Value An Econometric Analysis at the Firm Level
- M. HOUDEBINE A. TOPIOL-BENSAÏD G 9706 Les entreprises internationales en France : une analyse à partir de données individuelles
- M. HOUDEBINE G 9707 Polarisation des activités et spécialisation des départements en France
- E. DUGUET N. GREENAN G 9708 Le biais technologique : une analyse sur données individuelles
- G 9709 J.L. BRILLET Analyzing a small French ECM Model
- G 9710 J.L. BRILLET Formalizing the transition process: scenarios for capital accumulation
- G 9711 G. FORGEOT - J. GAUTIÉ Insertion professionnelle des jeunes et processus de déclassement
- G 9712 E. DUBOIS High Real Interest Rates: the Consequence of a Saving Investment Disequilibrium or of an insufficient Credibility of Monetary Authorities?
- Bilan des activités de la Direction des Études G 9713 et Synthèses Économiques - 1996

- E LEOUILLER G 9714 Does the French Consumer Price Index Overstate Inflation?
- G 9715 X BONNET Peut-on mettre en évidence les rigidités à la baisse des salaires nominaux ? Une étude sur quelques grands pays de l'OCDE
- N JUNG F RUPPRECHT G 9716 Productivité de la recherche et rendements d'échelle dans le secteur pharmaceutique francais
- G 9717 E DUGUET - I KABLA Appropriation strategy and the motivations to use the patent system in France - An econometric analysis at the firm level
- G 9718 L.P. PELÉ - P. RALLE Âge de la retraite : les aspects incitatifs du régime général
- G 9719 ZHANG Yingxiang - SONG Xueging Lexique macroéconomique français-chinois. chinois-francais
- M. HOUDEBINE J.L. SCHNEIDER G 9720 Mesurer l'influence de la fiscalité sur la localisation des entreprises
- G 9721 A. MOUROUGANE Crédibilité, indépendance et politique monétaire Une revue de la littérature
- G 9722 P. AUGERAUD - L. BRIOT Les données comptables d'entreprises Le système intermédiaire d'entreprises Passage des données individuelles aux données sectorielles
- G 9723 P. AUGERAUD - J.E. CHAPRON Using Business Accounts for Compiling National Accounts: the French Experience
- P. AUGERAUD G 9724 Les comptes d'entreprise par activités - Le passage aux comptes - De la comptabilité d'entreprise à la comptabilité nationale - A paraître
- H. MICHAUDON C. PRIGENT G 9801 Présentation du modèle AMADEUS
- G 9802 J. ACCARDO Une étude de comptabilité générationnelle pour la France en 1996
- G 9803 X BONNET - S DUCHÊNE Apports et limites de la modélisation « Real Business Cycles »
- G 9804 C. BARLET - C. DUGUET -D ENCAQUA - J PRADEL The Commercial Success of Innovations An econometric analysis at the firm level in French manufacturing
- G 9805 P. CAHUC - Ch. GIANELLA -D. GOUX - A. ZILBERBERG Equalizing Wage Differences and Bargaining Power - Evidence form a Panel of French Firms
- G 9806 J. ACCARDO - M. JLASSI La productivité globale des facteurs entre 1975 et 1996

- Bilan des activités de la Direction des Études et G 9807 Synthèses Économiques - 1997
- G 9808 A. MOUROUGANE Can a Conservative Governor Conduct an Accomodative Monetary Policy?
- X. BONNET E. DUBOIS L. FAUVET G 9809 Asymétrie des inflations relatives et menus costs : tests sur l'inflation francaise
- G 9810 E. DUGUET - N. IUNG Sales and Advertising with Spillovers at the firm level: Estimation of a Dynamic Structural Model on Panel Data
- G 9811 J.P. BERTHIER Congestion urbaine : un modèle de trafic de pointe à courbe débit-vitesse et demande élastique
- G 9812 C. PRIGENT La part des salaires dans la valeur aioutée : une approche macroéconomique
- G 9813 A.Th. AERTS L'évolution de la part des salaires dans la valeur ajoutée en France reflète-t-elle les évolutions individuelles sur la période 1979-1994?
- G 9814 B. SALANIÉ Guide pratique des séries non-stationnaires
- G 9901 S. DUCHÊNE - A. JACQUOT Une croissance plus riche en emplois depuis le début de la décennie ? Une analyse en comparaison internationale
- G 9902 Ch. COLIN Modélisation des carrières dans Destinie
- G 9903 Ch COLIN Évolution de la dispersion des salaires : un essai de prospective par microsimulation
- G 9904 B. CREPON - N. IUNG Innovation, emploi et performances
- G 9905 B. CREPON - Ch. GIANELLA Wages inequalities in France 1969-1992 An application of quantile regression techniques
- C. BONNET R. MAHIEU G 9906 Microsimulation techniques applied to intergenerational transfers - Pensions in a dynamic framework: the case of France
- G 9907 F ROSENWALD L'impact des contraintes financières dans la décision d'investissement
- G 9908 Bilan des activités de la DESE - 1998
- G 9909 J.P. ZOYEM Contrat d'insertion et sortie du RMI Évaluation des effets d'une politique sociale
- G 9910 Ch. COLIN - FI. LEGROS - R. MAHIEU Bilans contributifs comparés des régimes de retraite du secteur privé et de la fonction publique
- G 9911 G. LAROQUE - B. SALANIÉ Une décomposition du non-emploi en France
- G 9912 B. SALANIÉ Une maquette analytique de long terme du marché du travail
- G 9912 Ch GIANELLA

- Une estimation de l'élasticité de l'emploi peu qualifié à son coût G 9913 Division « Redistribution et Politiques Sociales » Le modèle de microsimulation dynamique DESTINIE
- G 9914 E. DUGUET Macro-commandes SAS pour l'économétrie des panels et des variables qualitatives
- G 9915 R. DUHAUTOIS Évolution des flux d'emplois en France entre 1990 et 1996 : une étude empirique à partir du fichier des bénéfices réels normaux (BRN)
- G 9916 J.Y. FOURNIER Extraction du cvcle des affaires : la méthode de Baxter et King
- G 9917 B. CRÉPON - R. DESPLATZ - J. MAIRESSE Estimating price cost margins, scale economies and workers' bargaining power at the firm level
- G 9918 Ch. GIANELLA - Ph. LAGARDE Productivity of hours in the aggregate production function: an evaluation on a panel of French firms from the manufacturing sector
- G 9919 S. AUDRIC - P. GIVORD - C. PROST Évolution de l'emploi et des coûts par qualification entre 1982 et 1996
- G 2000/01 R MAHIELL Les déterminants des dépenses de santé : une approche macroéconomique
- G 2000/02 C. ALLARD-PRIGENT H. GUILMEAU -A. QUINET The real exchange rate as the relative price of nontrables in terms of tradables: theoretical investigation and empirical study on French data
- J.-Y. FOURNIER G 2000/03 L'approximation du filtre passe-bande proposée par Christiano et Fitzgerald
- G 2000/04 Bilan des activités de la DESE - 1999
- B CREPON F ROSENWALD G 2000/05 Investissement et contraintes de financement : le poids du cycle Une estimation sur données françaises
- G 2000/06 A. FLIPO Les comportements matrimoniaux de fait
- G 2000/07 R. MAHIEU - B. SÉDILLOT Microsimulations of the retirement decision: a supply side approach
- G 2000/08 C. AUDENIS - C. PROST Déficit conioncturel : une prise en compte des conjonctures passées
- G 2000/09 R. MAHIEU - B. SÉDILLOT Équivalent patrimonial de la rente et souscription de retraite complémentaire
- G 2000/10 R. DUHAUTOIS Ralentissement de l'investissement : petites ou grandes entreprises ? industrie ou tertiaire ?
- G 2000/11 G. LAROQUE - B. SALANIÉ Temps partiel féminin et incitations financières à l'emploi
- Ch. GIANELLA G2000/12 Local unemployment and wages

Bis

- G2000/13 B. CREPON Th. HECKEL - Informatisation en France : une évaluation à partir de données individuelles - Computerization in France: an evaluation based on individual company data
- G2001/01 F. LEQUILLER - La nouvelle économie et la mesure de la croissance du PIB - The new economy and the measure ment of GDP growth
- G2001/02 S. AUDRIC La reprise de la croissance de l'emploi profite-telle aussi aux non-diplômés ?
- G2001/03 I. BRAUN-LEMAIRE Évolution et répartition du surplus de productivité
- G2001/04 A. BEAUDU Th. HECKEL Le canal du crédit fonctionne-t-il en Europe ? Une étude de l'hétérogénéité des comportements d'investissement à partir de données de bilan agrégées
- G2001/05 C. AUDENIS P. BISCOURP -N. FOURCADE - O. LOISEL Testing the augmented Solow growth model: An empirical reassessment using panel data
- G2001/06 R. MAHIEU B. SÉDILLOT Départ à la retraite, irréversibilité et incertitude
- G2001/07 Bilan des activités de la DESE 2000
- G2001/08 J. Ph. GAUDEMET Les dispositifs d'acquisition à titre facultatif d'annuités viagères de retraite
- G2001/09 B. CRÉPON Ch. GIANELLA Fiscalité, coût d'usage du capital et demande de facteurs : une analyse sur données individuelles
- G2001/10 B. CRÉPON R. DESPLATZ Évaluation des effets des dispositifs d'allégements de charges sociales sur les bas salaires
- G2001/11 J.-Y. FOURNIER Comparaison des salaires des secteurs public et privé
- G2001/12 J.-P. BERTHIER C. JAULENT R. CONVENEVOLE - S. PISANI Une méthodologie de comparaison entre consommations intermédiaires de source fiscale et de comptabilité nationale
- G2001/13 P. BISCOURP Ch. GIANELLA Substitution and complementarity between capital, skilled and less skilled workers: an analysis at the firm level in the French manufacturing industry
- G2001/14 I. ROBERT-BOBEE Modelling demographic behaviours in the French microsimulation model Destinie: An analysis of future change in completed fertility
- G2001/15 J.-P. ZOYEM Diagnostic sur la pauvreté et calendrier de revenus : le cas du "Panel européen des ménages »
- G2001/16 J.-Y. FOURNIER P. GIVORD La réduction des taux d'activité aux âges extrêmes, une spécificité française ?

- C. AUDENIS P. BISCOURP N. RIEDINGER G2001/17 Existe-t-il une asymétrie dans la transmission du prix du brut aux prix des carburants ? F. MAGNIEN - J.-L. TAVERNIER - D. THESMAR G2002/01 Les statistiques internationales de PIB par habitant en standard de pouvoir d'achat : une analyse des résultats G2002/02 Bilan des activités de la DESE - 2001 G2002/03 B. SÉDILLOT - E. WALRAET La cessation d'activité au sein des couples : y a-til interdépendance des choix ? G2002/04 G. BRILHAULT - Rétropolation des séries de FBCF et calcul du capital fixe en SEC-95 dans les comptes nationaux francais - Retropolation of the investment series (GFCF) and estimation of fixed capital stocks on the ESA-95 basis for the French balance sheets G2002/05 P. BISCOURP - B. CRÉPON - T. HECKEL - N. RIEDINGER How do firms respond to cheaper computers? Microeconometric evidence for France based on a production function approach G2002/06 C. AUDENIS - J. DEROYON - N. FOURCADE L'impact des nouvelles technologies de l'information et de la communication sur l'économie française - un bouclage macroéconomique
- G2002/07 J. BARDAJI B. SÉDILLOT E. WALRAET Évaluation de trois réformes du Régime Général d'assurance vieillesse à l'aide du modèle de microsimulation DESTINIE
- G2002/08 J.-P. BERTHIER Réflexions sur les différentes notions de volume dans les comptes nationaux : comptes aux prix d'une année fixe ou aux prix de l'année précédente, séries chaînées
- G2002/09 F. HILD Les soldes d'opinion résument-ils au mieux les réponses des entreprises aux enquêtes de conjoncture ?
- G2002/10 I. ROBERT-BOBÉE Les comportements démographiques dans le modèle de microsimulation Destinie - Une comparaison des estimations issues des enquêtes Jeunes et Carrières 1997 et Histoire Familiale 1999
- G2002/11 J.-P. ZOYEM La dynamique des bas revenus : une analyse des entrées-sorties de pauvreté
- G2002/12 F. HILD Prévisions d'inflation pour la France
- G2002/13 M. LECLAIR Réduction du temps de travail et tensions sur les facteurs de production
- G2002/14 E. WALRAET A. VINCENT - Analyse de la redistribution intragénérationnelle dans le système de retraite des salariés du privé -Une approche par microsimulation - Intragenerational distributional analysis in the french private sector pension scheme - A microsimulation approach

- G2002/15 P. CHONE D. LE BLANC I. ROBERT-BOBEE Offre de travail féminine et garde des jeunes enfants
- G2002/16 F. MAUREL S. GREGOIR Les indices de compétitivité des pays : interprétation et limites
- G2003/01 N. RIEDINGER E.HAUVY Le coût de dépollution atmosphérique pour les entreprises françaises : Une estimation à partir de données individuelles
- G2003/02 P. BISCOURP et F. KRAMARZ Création d'emplois, destruction d'emplois et internationalisation des entreprises industrielles françaises : une analyse sur la période 1986-1992
- G2003/03 Bilan des activités de la DESE 2002
- G2003/04 P.-O. BEFFY J. DEROYON -N. FOURCADE - S. GREGOIR - N. LAÏB -B. MONFORT Évolutions démographiques et croissance : une projection macro-économique à l'horizon 2020
- G2003/05 P. AUBERT La situation des salariés de plus de cinquante ans dans le secteur privé
- G2003/06 P. AUBERT B. CRÉPON Age, salaire et productivité La productivité des salariés décline-t-elle en fin de carrière ?
- G2003/07 H. BARON P.O. BEFFY N. FOURCADE R. MAHIEU Le ralentissement de la productivité du travail au cours des années 1990
- G2003/08 P.-O. BEFFY B. MONFORT Patrimoine des ménages, dynamique d'allocation et comportement de consommation
- G2003/09 P. BISCOURP N. FOURCADE Peut-on mettre en évidence l'existence de rigidités à la baisse des salaires à partir de données individuelles ? Le cas de la France à la fin des années 90
- G2003/10 M. LECLAIR P. PETIT Présence syndicale dans les firmes : quel impact sur les inégalités salariales entre les hommes et les femmes ?
- G2003/11 P.-O. BEFFY X. BONNET M. DARRACQ-PARIES - B. MONFORT MZE: a small macro-model for the euro area
- G2004/01 P. AUBERT M. LECLAIR La compétitivité exprimée dans les enquêtes trimestrielles sur la situation et les perspectives dans l'industrie
- G2004/02 M. DUÉE C. REBILLARD La dépendance des personnes âgées : une projection à long terme
- G2004/03 S. RASPILLER N. RIEDINGER Régulation environnementale et choix de localisation des groupes français
- G2004/04 A. NABOULET S. RASPILLER Les déterminants de la décision d'investir : une approche par les perceptions subjectives des firmes

- La déclaration des enfants par les couples non mariés est-elle fiscalement optimale ? G2004/06 M DUÉE L'impact du chômage des parents sur le devenir scolaire des enfants P. AUBERT - E. CAROLI - M. ROGER G2004/07 New Technologies. Workplace Organisation and the Age Structure of the Workforce: Firm-Level Evidence E. DUGUET - C. LELARGE G2004/08 Les brevets accroissent-ils les incitations privées à innover ? Un examen microéconométrique G2004/09 S. RASPILLER - P. SILLARD Affiliating versus Subcontracting: the Case of Multinationals G2004/10 J. BOISSINOT - C. L'ANGEVIN - B. MONFORT Public Debt Sustainability: Some Results on the French Case G2004/11 S. ANANIAN - P. AUBERT Travailleurs âgés, nouvelles technologies et changements organisationnels : un réexamen à partir de l'enquête « REPONSE » G2004/12 X. BONNET - H. PONCET Structures de revenus et propensions différentes à consommer - Vers une équation de consommation des ménages plus robuste en prévision pour la France G2004/13 C. PICART Évaluer la rentabilité des sociétés non financières J. BARDAJI - B. SÉDILLOT - E. WALRAET G2004/14 Les retraites du secteur public : projections à l'horizon 2040 à l'aide du modèle de microsimulation DESTINIE S. BUFFETEAU - P. GODEFROY G2005/01 Conditions de départ en retraite selon l'âge de fin d'études : analyse prospective pour les générations 1945 à 1974
- G2005/02 C. AFSA S. BUFFETEAU L'évolution de l'activité féminine en France : une approche par oseudo-panel
- G2005/03 P. AUBERT P. SILLARD Délocalisations et réductions d'effectifs dans l'industrie française
- G2005/04 M. LECLAIR S. ROUX Mesure et utilisation des emplois instables dans les entreprises
- G2005/05 C. L'ANGEVIN S. SERRAVALLE Performances à l'exportation de la France et de l'Allemagne - Une analyse par secteur et destination géographique
- G2005/06 Bilan des activités de la Direction des Études et Synthèses Économiques - 2004
- G2005/07 S. RASPILLER La concurrence fiscale : principaux enseignements de l'analyse économique
- G2005/08 C. L'ANGEVIN N. LAÏB Éducation et croissance en France et dans un panel de 21 pays de l'OCDE
- G2005/09 N. FERRARI Prévoir l'investissement des entreprises

G2004/05

N. RAGACHE

Un indicateur des révisions dans l'enquête de conioncture sur les investissements dans l'industrie.

- G2005/10 P.-O. BEFFY - C. L'ANGEVIN Chômage et boucle prix-salaires : apport d'un modèle « qualifiés/peu qualifiés »
- G2005/11 B. HEITZ A two-states Markov-switching model of inflation in France and the USA: credible target VS inflation spiral
- O. BIAU H. ERKEL-ROUSSE N. FERRARI G2005/12 Réponses individuelles aux enquêtes de conjoncture et prévision macroéconomiques : Exemple de la prévision de la production manufacturière
- P. AUBERT D. BLANCHET D. BLAU G2005/13 The labour market after age 50: some elements of a Franco-American comparison
- G2005/14 D. BLANCHET - T. DEBRAND -P. DOURGNON - P. POLLET L'enquête SHARE : présentation et premiers résultats de l'édition française
- G2005/15 M. DUÉE La modélisation des comportements démographiques dans le modèle de microsimulation
- G2005/16 H. RAOUI - S. ROUX Étude de simulation sur la participation versée aux salariés par les entreprises
- C. BONNET S. BUFFETEAU P. GODEFROY G2006/01 Disparités de retraite de droit direct entre hommes et femmes : quelles évolutions ?
- C. PICART G2006/02 Les gazelles en France
- P. AUBERT B. CRÉPON P. ZAMORA G2006/03 Le rendement apparent de la formation continue dans les entreprises : effets sur la productivité et les salaires
- J.-F. OUVRARD R. RATHELOT G2006/04 Demographic change and unemployment: what do macroeconometric models predict?
- G2006/05 D. BLANCHET - J.-F. OUVRARD Indicateurs d'engagements implicites des systèmes de retraite : chiffrages, propriétés analytiques et réactions à des chocs démographiques types
- G. BIAU O. BIAU L. ROUVIERE G2006/06 Nonparametric Forecasting of the Manufacturing Output Growth with Firm-level Survey Data
- G2006/07 C AESA - P GIVORD Le rôle des conditions de travail dans les absences pour maladie
- G2006/08 P. SILLARD - C. L'ANGEVIN - S. SERRAVALLE Performances comparées à l'exportation de la France et de ses principaux partenaires Une analyse structurelle sur 12 ans
- X. BOUTIN S. QUANTIN G2006/09 Une méthodologie d'évaluation comptable du coût du capital des entreprises françaises : 1984-2002

- G2006/10 C. AFSA L'estimation d'un coût implicite de la pénibilité du travail chez les travailleurs âgés G2006/11 C. I.FLARGE Les entreprises (industrielles) francaises sontelles à la frontière technologique ? O. BIAU - N. FERRARI G2006/12 Théorie de l'opinion Faut-il pondérer les réponses individuelles ? G2006/13 A. KOUBI - S. ROUX Une réinterprétation de la relation entre productivité et inégalités salariales dans les . entreprises G2006/14 R. RATHELOT - P. SILLARD The impact of local taxes on plants location decision. G2006/15 L. GONZALEZ - C. PICART Diversification, recentrage et poids des activités de support dans les groupes (1993-2000) G2007/01 D. SRAER Allègements de cotisations patronales et dynamique salariale G2007/02 V. ALBOUY - L. LEQUIEN Les rendements non monétaires de l'éducation : le cas de la santé G2007/03 D. BLANCHET - T. DEBRAND Aspiration à la retraite, santé et satisfaction au travail : une comparaison européenne M. BARLET - L. CRUSSON G2007/04 Quel impact des variations du prix du pétrole sur la croissance française ? C PICART G2007/05 Flux d'emploi et de main-d'œuvre en France : un réexamen G2007/06 V. ALBOUY - C. TAVAN Massification et démocratisation de
- l'enseignement supérieur en France G2007/07 T. LE BARBANCHON The Changing response to oil price shocks in France: a DSGE type approach
- T. CHANEY D. SRAER D. THESMAR G2007/08 Collateral Value and Corporate Investment Evidence from the French Real Estate Market
- G2007/09 J. BOISSINOT Consumption over the Life Cycle: Facts for France
- G2007/10 C. AFSA Interpréter les variables de satisfaction : l'exemple de la durée du travail
- R. RATHELOT P. SILLARD G2007/11 Zones Franches Urbaines: quels effets sur l'emploi salarié les créations et d'établissements ?
- G2007/12 V. ALBOUY - B. CRÉPON Aléa moral en santé : une évaluation dans le cadre du modèle causal de Rubin
- G2008/01 C. PICART Les PME françaises : rentables mais peu dynamiques

The Effects of Retail Regulations on Prices Evidence form the Loi Galland G2008/03 Y. BARBESOL - A. BRIANT Économies d'agglomération et productivité des entreprises : estimation sur données individuelles francaises D BLANCHET - E LE GALLO G2008/04 Les projections démographiques : principaux mécanismes et retour sur l'expérience française D BLANCHET - E TOUTLEMONDE G2008/05 Évolutions démographiques et déformation du cycle de vie active : quelles relations ?

G2008/02

P. BISCOURP - X. BOUTIN - T. VERGÉ

- G2008/06 M. BARLET - D. BLANCHET - L. CRUSSON Internationalisation et flux d'emplois : que dit une approche comptable ?
- G2008/07 C. LELARGE - D. SRAER - D. THESMAR Entrepreneurship and Credit Constraints -Evidence from a French Loan Guarantee Program
- G2008/08 X. BOUTIN - L. JANIN Are Prices Really Affected by Mergers?
- G2008/09 M. BARLET - A. BRIANT - L. CRUSSON Concentration géographique dans l'industrie manufacturière et dans les services en France : une approche par un indicateur en continu
- G2008/10 M. BEFFY - É. COUDIN - R. RATHELOT Who is confronted to insecure labor market histories? Some evidence based on the French labor market transition
- M. ROGER E. WALRAET G2008/11 Social Security and Well-Being of the Elderly: the Case of France
- C. AFSA G2008/12 Analyser les composantes du bien-être et de son évolution Une approche empirique sur données individuelles
- G2008/13 M. BARLET - D. BLANCHET -T. LE BARBANCHON Microsimuler le marché du travail : un prototype
- G2009/01 P-A PIONNIER Le partage de la valeur ajoutée en France. 1949-2007
- G2009/02 Laurent CLAVEL - Christelle MINODIER A Monthly Indicator of the French Business Climate
- G2009/03 H. ERKEL-ROUSSE - C. MINODIER Do Business Tendency Surveys in Industry and Services Help in Forecasting GDP Growth? A Real-Time Analysis on French Data
- P. GIVORD L. WILNER G2009/04 Les contrats temporaires : trappe ou marchepied vers l'emploi stable ?
- G2009/05 G. LALANNE - P.-A. PIONNIER - O. SIMON Le partage des fruits de la croissance de 1950 à 2008 : une approche par les comptes de surplus
- G2009/06 L. DAVEZIES - X. D'HAULTFOEUILLE Faut-il pondérer ?... Ou l'éternelle question de l'économètre confronté à des données d'enquête
- S. QUANTIN S. RASPILLER S. SERRAVALLE G2009/07 Commerce intragroupe, fiscalité et prix de transferts : une analyse sur données francaises G2009/08 M. CLERC - V. MARCUS Élasticités-prix des consommations énergétiques des ménages G2009/09 G. LALANNE - E. POULIQUEN - O. SIMON Prix du pétrole et croissance potentielle à long terme G2009/10 D. BLANCHET - J. LE CACHEUX - V. MARCUS Adjusted net savings and other approaches to sustainability: some theoretical background G2009/11 V. BELLAMY - G. CONSALES - M. FESSEAU -S. LE LAIDIER - É. RAYNAUD Une décomposition du compte des ménages de la comptabilité nationale par catégorie de ménage en 2003 G2009/12 J. BARDAJI - F. TALLET Detecting Economic Regimes in France: a Qualitative Markov-Switching Indicator Using Mixed Frequency Data R. AEBERHARDT - D. FOUGÈRE -G2009/13 R. RATHELOT Discrimination à l'embauche : comment exploiter les procédures de testina? G2009/14 Y. BARBESOL - P. GIVORD - S. QUANTIN Partage de la valeur ajoutée, approche par données microéconomiques G2009/15 I. BUONO - G. LALANNE The Effect of the Uruguay round on the Intensive and Extensive Margins of Trade G2010/01 C. MINODIER Avantages comparés des séries des premières valeurs publiées et des séries des valeurs révisées - Un exercice de prévision en temps réel de la croissance trimestrielle du PIB en France G2010/02 V. ALBOUY - L. DAVEZIES - T. DEBRAND Health Expenditure Models: a Comparison of Five Specifications using Panel Data C. KLEIN - O. SIMON G2010/03 Le modèle MÉSANGE réestimé en base 2000 Tome 1 - Version avec volumes à prix constants G2010/04 M.-É. CLERC - É. COUDIN L'IPC, miroir de l'évolution du coût de la vie en France? Ce qu'apporte l'analyse des courbes d'Engel G2010/05 N. CECI-RENAUD - P.-A. CHEVALIER Les seuils de 10, 20 et 50 salariés : impact sur la taille des entreprises françaises G2010/06 R AFBERHARDT - J POUGET National Origin Differences in Wages and Hierarchical Positions - Evidence on French Full-Time Male Workers from a matched Employer-Employee Dataset G2010/07 S. BLASCO - P. GIVORD Les trajectoires professionnelles en début de vie active : quel impact des contrats temporaires ? P. GIVORD G2010/08
  - Méthodes économétriques pour l'évaluation de politiques publiques

- ix
- P.-Y. CABANNES V. LAPÈGUE -G2010/09 E. POULIQUEN - M. BEFFY - M. GAINI Quelle croissance de moven terme après la crise ? G2010/10 I. BUONO - G. LALANNE La réaction des entreprises françaises à la baisse des tarifs douaniers étrangers R RATHELOT - P SILLARD G2010/11 L'apport des méthodes à novaux pour mesurer la concentration géographique - Application à la concentration des immigrés en France de 1968 à 1999 M. BARATON - M. BEFFY - D. FOUGÈRE G2010/12 Une évaluation de l'effet de la réforme de 2003 sur les départs en retraite - Le cas des enseignants du second degré public D. BLANCHET - S. BUFFETEAU - E. CRENNER G2010/13 S. LE MINEZ Le modèle de microsimulation Destinie 2 : principales caractéristiques et premiers résultats G2010/14 D. BLANCHET - E. CRENNER Le bloc retraites du modèle Destinie 2 : quide de l'utilisateur G2010/15 M. BARLET - L. CRUSSON - S. DUPUCH -F. PUECH Des services échangés aux services échangeables : une application sur données françaises G2010/16 M. BEFFY - T. KAMIONKA Public-private wage gaps: is civil-servant human capital sector-specific? G2010/17 P.-Y. CABANNES - H. ERKEL-ROUSSE G. LALANNE - O. MONSO - E. POULIQUEN Le modèle Mésange réestimé en base 2000 Tome 2 - Version avec volumes à prix chaînés G2010/18 R. AEBERHARDT - L. DAVEZIES Conditional Logit with one Binary Covariate: Link between the Static and Dynamic Cases G2011/01 T. LE BARBANCHON - B. OURLIAC - O. SIMON Les marchés du travail français et américain face aux chocs conjoncturels des années 1986 à 2007 : une modélisation DSGE C. MARBOT G2011/02 Une évaluation de la réduction d'impôt pour l'emploi de salariés à domicile G2011/03 L. DAVEZIES Modèles à effets fixes, à effets aléatoires, modèles mixtes ou multi-niveaux : propriétés et mises en œuvre des modélisations de l'hétérogénéité dans le cas de données groupées M. ROGER - M. WASMER G2011/04 Heterogeneity matters: labour productivity differentiated by age and skills G2011/05 J-C BRICONGNE - J-M FOURNIER V. LAPÈGUE - O. MONSO De la crise financière à la crise économique L'impact des perturbations financières de 2007 et 2008 sur la croissance de sept pays industrialisés P. CHARNOZ - É. COUDIN - M. GAINI G2011/06 Wage inequalities in France 1976-2004: a quantile regression analysis
- G2011/07 M. CLERC M. GAINI D. BLANCHET Recommendations of the Stiglitz-Sen-Fitoussi Report: A few illustrations
  - G2011/08 M. BACHELET M. BEFFY D. BLANCHET Projeter l'impact des réformes des retraites sur l'activité des 55 ans et plus : une comparaison de trois modèles
  - G2011/09 C. LOUVOT-RUNAVOT L'évaluation de l'activité dissimulée des entreprises sur la base des contrôles fiscaux et son insertion dans les comptes nationaux
  - G2011/10 A. SCHREIBER A. VICARD La tertiarisation de l'économie française et le ralentissement de la productivité entre 1978 et 2008
  - G2011/11 M.-É. CLERC O. MONSO E. POULIQUEN Les inégalités entre générations depuis le babyboom
  - G2011/12 C. MARBOT et D. ROY Évaluation de la transformation de la réduction d'impôt en crédit d'impôt pour l'emploi de salariés à domicile en 2007
  - G2011/13 P. GIVORD R. RATHELOT P. SILLARD Place-based tax exemptions and displacement effects: An evaluation of the Zones Franches Urbaines program
  - G2011/14 X. D'HAULTFOEUILLE P. GIVORD -X. BOUTIN The Environmental Effect of Green Taxation: the Case of the French "Bonus/Malus"
  - G2011/15 M. BARLET M. CLERC M. GARNEO -V. LAPÉGUE - V. MARCUS La nouvelle version du modèle MZE, modèle macroéconométrique pour la zone euro
  - G2011/16 R. AEBERHARDT I. BUONO H. FADINGER Learning, Incomplete Contracts and Export Dynamics: theory and Evidence form French Firms
  - G2011/17 C. KERDRAIN V. LAPÈGUE Restrictive Fiscal Policies in Europe: What are the Likely Effects?
  - G2012/01 P. GIVORD S. QUANTIN C. TREVIEN A Long-Term Evaluation of the First Generation of the French Urban Enterprise Zones
  - G2012/02 N. CECI-RENAUD V. COTTET Politique salariale et performance des entreprises
  - G2012/03 P. FÉVRIER L. WILNER Do Consumers Correctly Expect Price Reductions? Testing Dynamic Behavior
  - G2012/04 M. GAINI A. LEDUC A. VICARD School as a shelter? School leaving-age and the business cycle in France
  - G2012/05 M. GAINI A. LEDUC A. VICARD A scarred generation? French evidence on young people entering into a tough labour market
  - G2012/06 P. AUBERT M. BACHELET Disparités de montant de pension et redistribution dans le système de retraite français
  - G2012/07 R. AEBERHARDT P GIVORD C. MARBOT Spillover Effect of the Minimum Wage in France: An Unconditional Quantile Regression Approach

- G2012/08 A. EIDELMAN F. LANGUMIER A. VICARD Prélèvements obligatoires reposant sur les ménages : des canaux redistributifs différents en 1990 et 2010
- G2012/09 O. BARGAIN A. VICARD Le RMI et son successeur le RSA découragentils certains jeunes de travailler ? Une analyse sur les jeunes autour de 25 ans
- G2012/10 C. MARBOT D. ROY Projections du coût de l'APA et des caractéristiques de ses bénéficiaires à l'horizon 2040 à l'aide du modèle Destinie
- G2012/11 A. MAUROUX Le crédit d'impôt dédié au développement durable : une évaluation économétrique
- G2012/12 V. COTTET S. QUANTIN V. RÉGNIER Coût du travail et allègements de charges : une estimation au niveau établissement de 1996 à 2008
- G2012/13 X. D'HAULTFOEUILLE, P. FEVRIER et L. WILNER Demand Estimation in the Presence of Revenue Management
- G2012/14 D. BLANCHET et S. LE MINEZ Joint macro/micro evaluations of accrued-to-date pension liabilities: an application to French reforms
- G2013/01- T. DEROYON A. MONTAUT et P-A PIONNIER F1301 Utilisation rétrospective de l'enquête Emploi à une fréquence mensuelle : apport d'une modélisation espace-état
- G2013/02- C. TRÉVIEN
- F1302 Habiter en HLM: quel avantage monétaire et quel impact sur les conditions de logement ?
- G2013/03 A. POISSONNIER Temporal disaggregation of stock variables - The Chow-Lin method extended to dynamic models
- G2013/04 P. GIVORD C. MARBOT Does the cost of child care affect female labor market participation? An evaluation of a French reform of childcare subsidies
- G2013/05 G. LAME M. LEQUIEN P.-A. PIONNIER Interpretation and limits of sustainability tests in public finance
- G2013/06 C. BELLEGO V. DORTET-BERNADET La participation aux pôles de compétitivité : quelle incidence sur les dépenses de R&D et l'activité des PME et ETI ?
- G2013/07 P-Y. CABANNES A.MONTAUT -P-A. PIONNIER Évaluer la productivité globale des facteurs en France : l'apport d'une mesure de la qualité du capital et du travail
- G2013/08 R. AEBERHARDT C. MARBOT Evolution of Instability on the French Labour Market During the Last Thirty Years
- G2013/09 J-B. BERNARD G. CLÉAUD Oil price: the nature of the shocks and the impact on the French economy
- G2013/10 G. LAME Was there a « Greenspan Conundrum » in the Euro area?

- G2013/11 P. CHONÉ F. EVAIN L. WILNER E. YILMAZ Introducing activity-based payment in the hospital industry : Evidence from French data
- G2013/12 C. GRISLAIN-LETRÉMY Natural Disasters: Exposure and Underinsurance